

21/01099

FFI-RAPPORT

Efficacy of face masks as source control and respiratory protection against transmission of SARS-CoV-2

Marius Dybwad Else Marie Fykse Thor Gjesdal Agnieszka Anna Gorzkowska-Sobas Anders Helgeland Espen Åkervik

Efficacy of face masks as source control and respiratory protection against transmission of SARS-CoV-2

Marius Dybwad Else Marie Fykse Thor Gjesdal Agnieszka Anna Gorzkowska-Sobas Anders Helgeland Espen Åkervik

Norwegian Defence Research Establishment (FFI)

25 June 2021

Keywords

Covid-19 SARS-CoV-2 Virus Beskyttelse Aerosoler Mikrobiologi

FFI report 21/01099

Project number 564501

Electronic ISBN 978-82-464-3361-5

Approvers Janet Martha Blatny, *Research Director*

The document is electronically approved and therefore has no handwritten signature.

Copyright

© Norwegian Defence Research Establishment (FFI). The publication may be freely cited where the source is acknowledged.

Summary

During the COVID-19 pandemic, medical and community face masks have been used extensively by health care workers and the general public to reduce transmission of the SARS-CoV-2 virus.

This report reviews the available literature on the efficacy of medical and community face masks, both as a means for source control and respiratory protection. The literature spans a multitude of scientific disciplines. The review has, however, mainly focused on aerosol physics, aerobiology, microbiology and occupational health and safety.

The literature shows that medical and community face masks remove almost all large and medium-sized respiratory droplets from the exhaled air. To some extent, they also reduce the number of emitted small droplets. All respiratory activities generate turbulent jet plumes in the direction of exhalation. These directed jet plumes are able to augment the travel distance of droplets of all sizes and lead to localized regions with elevated droplet concentrations. An important function of face coverings, such as medical and community face masks, is their ability to obstruct these jets, and thereby diffuse the droplet concentration. Due to the combined effect of the above factors, medical and community face masks appear to be highly efficient when used as a means for source control.

On the other hand, medical and community face masks do not typically fulfill the performance requirements that apply to respiratory protection devices. The reason for this is twofold. First, medical and community face masks are loose-fit devices prone to gap leakages around the perimeter. Therefore, a substantial fraction of the respiratory droplets may bypass the filtering layer of the mask. Second, the filtration efficiency for very small droplets is highly variable for medical and community face masks.

This finding is not surprising, since the intended use for medical face masks as a personal protective device is limited to direct transmission via splashes and sprays. The performance requirements and test methods applied to such devices, according to the relevant harmonized standards, do therefore not consider their function as a respiratory protection device. The role of very small droplets in the transmission of SARS-COV-2 is currently not well understood. Therefore, it is difficult to ascertain whether the strict performance requirements for respiratory protection devices are necessary to achieve a sufficient protection level. Some studies show that the use of medical face masks may offer some degree of respiratory protection to the wearer, in particular if additional measures such as double masking or knotting and tucking are used to improve the fit and reduce gap leakages.

Contents

Summary				
Co	ontents	5		5
1	Intro	ductior	n	7
2	Transmission routes for SARS-CoV-2			9
	2.1	SARS	S-CoV-2	11
		2.1.1	Environmental stability	12
		2.1.2	Infectious dose and viral load	13
	2.2	Trans	mission of SARS-CoV-2 by respiratory droplets	15
		2.2.1	Droplet generation	15
		2.2.2	Droplet transport	17
	2.3	Sourc	e control versus respiratory protection	19
3	Face	21		
	3.1	Basic	terms and definitions	22
		3.1.1	Filtration efficiency (FE)	22
		3.1.2	Fit factor (FF)	22
		3.1.3	Total Inward Leakage (TIL)	23
		3.1.4	Protection factor (PF)	23
		3.1.5	Pressure resistance	23
	3.2	Alternative methods for filtration efficiency testing		24
	3.3	The mechanics of face coverings		25
	3.4	Filterin	ng facepiece respirators (FFRs)	27
		3.4.1	Product categories	29
		3.4.2	Performance requirements and test methodology	29
	3.5	Medic	al face masks (MFMs)	30
		3.5.1	Product categories	31
		3.5.2	Performance requirements and test methodology	31
	3.6	Comm	nunity face masks (CFMs)	33
		3.6.1	Product categories	34
		3.6.2	Performance requirements and test methodology	34
	3.7	Faces	shields	35
		3.7.1	Product categories	35

		3.7.2	Performance requirements and test methodology	35
4	Filtra	ation eff	iciency of face masks	36
	4.1	Filtrati	on efficiency of medical face masks	37
	4.2	Filtrati	on efficiency of community face masks	38
	4.3	Answe	ers to key question in Chapter 4	40
5	Effic	acy of f	ace masks as source control against transmission of SARS-CoV-2	41
	5.1	Obstru	icting the respiratory jet	42
	5.2	Mask	fit and outward leakage	44
	5.3	Answe	ers to key questions in Chapter 5	48
6	Effic	acy of f	ace masks as respiratory protection against transmission of SARS-	
	CoV-2			50
	6.1	Mask	fit and inward leakage	50
	6.2	Answe	er to key questions in Chapter 6	55
7	The effects of long term use, humidity, washing and reuse of face masks			
	7.1	Answe	ers to key questions in Chapter 7	58
8	Disc	ussion	and conclusions	59
Re	feren	ces		62

1 Introduction

Coronavirus disease (COVID-19) is an infectious disease caused by a newly discovered coronavirus, severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2), previously not found in humans. The World Health Organization (WHO) declared COVID-19 a public health emergency of international concern (PHEIC) on January 30 2020 and a global pandemic on March 11 2020 [1]. As of June 2021 more than 179 million people have been infected and as many as 3.8 million people have died [2]. As a result, the health care systems in most countries have been under severe and persisting strain, including periods of shortage of critical human resources as well as medical and protective equipment. In order to reduce or avoid SARS-CoV-2 transmission, human disease and deaths, and collapse of organized health care services, governments across the world have instituted general mandatory use of face coverings for the general public in situations where close contact is unavoidable. Public masking is also recommended by organizations and institutions responsible for infectious disease control and prevention, among others the European Centre for Disease Prevention and Control (ECDC) [3], the U.S. Centers for Disease Control and Prevention (CDC) [4] and the WHO [5].

Health care workers (HCWs) have used various face coverings, including medical face masks (MFMs), filtering facepiece respirators (FFRs), powered air-purifying respirators (PAPRs) and face shields, to reduce SARS-CoV-2 transmission in the hospital environment by protecting HCWs and patients against virus-laden respiratory droplets and aerosols. The utter need to protect HCWs from infection have caused an increased demand for personal protective equipment, in particular respiratory protection devices (RPDs) such as FFRs and PAPRs. MFMs are classified as medical devices and not RPDs. According to the relevant standards their primary intended use is source control, i.e. to control and prevent emission of respiratory droplets from the wearer in order to protect others. MFMs are typically used by HCWs in the hospital environment to protect patients from respiratory tract-associated microorganisms emitted by HCWs during surgery and other medical procedures. As a personal protection device, the intended use for medical face masks is limited to direct transmission via splashes and sprays including large respiratory droplets. In the specific context of small and medium-sized respiratory droplets, the intended use for MFMs is limited to source control. The performance requirements and test methods applied to such devices according to the relevant standards do therefore not consider their function as a RPD.

During the COVID-19 pandemic, the use of MFMs has become widespread also among the general public. As a result, the use of these devices have surged, which at times have resulted in poor availability and a skyrocketing market prices [6, 7]. Both for RPDs and MFMs, which are typically single-use devices, there have also been concerns regarding the amount of plastics produced and utilized, and the resulting environmental impact [8-11]. Re-usable (washable) cloth masks, both commercial and home-made versions, have therefore gained popularity among the general public [5, 12]. To clearly separate such face coverings from MFMs, non-medical cloth masks for community use are therefore categorized and referred to as community face masks (CFMs) throughout the rest of this report.

MFMs and CFMs are loose-fitting devices with relatively low breathing resistance, which does not contribute significantly to a physical and psychological burden for the wearer. MFMs and CFMs are also more readily available than RPDs and have a simpler design and form that does not require fit testing. MFMs and CFMs can therefore be used by a much larger group of individuals, with different facial anthropomorphic dimensions, facial hair, etc., and better compliance can be expected given their easier donning and lower price. However, many questions have been and are still being asked regarding the widespread use of MFMs and CFMs during the COVID-19 pandemic and the actual benefit/efficacy that can be expected in terms of preventing and controlling SARS-CoV-2 transmission both inside the health care service and in the community. Several of these questions have so far been difficult to answer because of a lack of knowledge regarding COVID-19, and to some extent also viral respiratory diseases in general, concerning the benefit/efficacy of MFMs and CFMs in preventing and controlling SARS-CoV-2 transmission when used as a mean for source control and respiratory protection. Another challenge is the fact that expertise from a variety of different scientific disciplines, including infectious disease medicine, epidemiology, occupational health and safety, infection prevention and control, microbiology, aerobiology and aerosol physics, is needed, and which is seldom accessible within any single organization. This in turn demands a cross-disciplinary approach and coordination, which is of vital importance to fully grasp, reconcile and answer many of these questions and truly understand and communicate their implications. Although a massive global research effort has been launched since the start of the COVID-19 pandemic, much of the emerging scientific knowledge is still captured only in pre-print publications that have not yet been peer reviewed. Even the sheer amount of new information, which due to the inherent nature of scientific process consists of a complex blend of consistent and conflicting findings, makes it difficult even for the experts to navigate this landscape in search of tangible information and answers.

In this report, we address and critically discuss, and whenever possible also attempt to answer, some of the questions asked about face coverings based on a cross-disciplinary review of the available scientific literature. Our review is focused on aerosol physics, aerobiology, microbiology, and occupational health and safety, and we attempt to describe the current state-of-knowledge and scientific evidence concerning the benefit/efficacy of MFMs and CFMs in the specific context of; a) source control and b) personal respiratory protection, from different perspectives, including filtration efficiency of filter materials, mask fit and outward/inward leakage, and handling and (re-)use issues.

Briefly, Chapter 2 covers transmission of SARS-CoV-2 via respiratory droplets. Chapter 3 covers different types of face coverings, particularly MFMs and CFMs, and introduces important terms and definitions used to describe their properties, performance and test methodology. Chapter 4 covers the filtration efficiency of MFMs and CFMs. Chapters 5 and 6 cover the efficacy of MFMs and CFMs as source control and respiratory protection, respectively, against transmission of SARS-CoV-2. Chapter 7 covers topics related to long-term use, humidity, washing and reuse of CFMs. Finally, Chapter 8 rounds of the report with a discussion and conclusions.

2 Transmission routes for SARS-CoV-2

The transmission routes for SARS-CoV-2 has been a subject to much debate since the beginning of the pandemic [13-18]. Historically, three different transmission modes of viral respiratory infections have been distinguished: (i) contact transmission, (ii) droplet transmission, and (iii) airborne transmission. CDC has recently attempted to clarify the transmission routes relevant for SARS-CoV-2 [14], which are now referred to as (i) inhalation of virus, (ii) deposition of virus on exposed mucous membranes, and (iii) touching mucous membranes with soiled hands contaminated with virus. Note that there is not a strict one-to-one correspondence between the two sets of transmission route definitions. In particular, the inhalation route will comprise transmission both by small droplets and droplet nuclei, commonly referred to as aerosols, as well as inhalable medium-sized droplets.

The size distribution of respiratory droplets spans a continuous range of diameters from submicron to about 1 mm. After exhalation, evaporation will tend to shrink the droplets, but this will not significantly affect the span of the droplet size distribution. An operational classification into small, medium and large droplets can be obtained by considering their mechanical properties. In particular, the interplay of gravitational settling, inertia, and aerodynamic drag determines how they move in an airflow. Small droplets evaporate rapidly into droplet nuclei that have negligible settling velocity and inertia. As a consequence, they respond quickly to the aerodynamic drag imposed by the airflow and follow the air passively. In terms of indoor ventilation flows and respiratory flows, this may apply to droplets smaller than approximately 15 µm. In terms of pathogen transmission, droplets smaller than 5 µm have traditionally been termed aerosols, although this usage of the term aerosol is not consistent with the use in other disciplines [19]. Probably, the 5 μ m definition was linked to the so-called respirable fraction [20, 21], i.e. droplets small enough to penetrate the lower airways. Large droplets have substantial settling velocities, i.e tens of centimeter per second, slower evaporation rates, and their inertia is so large that they respond slowly to changes in the airflow, thus displaying ballistic behaviour. For indoor ventilation and respiratory flows, this may refer to droplets larger than approximately 100 μ m. In between these cases, we find medium-sized droplets with a mixed behavior. These droplets may readily follow the airflow despite a non-negligble settling velocity. According to ISO 7708:1995(E) [20], droplets smaller than 100 µm constitute the inhalable fraction. Throughout the rest of the report, and in accordance with Stadnytskyi et al. [22], we will use the above operational definition of small, medium and large droplets despite the fuzzy boundaries between the different classes.

There is evidence that the risk of infection is greatest within 2 m distance of an infected source [14]. Within such distances, it is difficult to distinguish between the different modes of transmission [23]. Recent evidence suggests that contact with contaminated surfaces (fomites) plays a less important role than previously assumed in transmission of SARS-CoV-2 [15-18]. WHO has also acknowledged that in addition to large droplets, transmission of SARS-CoV-2 via small droplets and droplet nuclei (aerosols) cannot be ruled out, especially in enclosed indoor settings [5, 24]. In particular, small droplets are frequently produced in hospital environments during aerosol generating procedures [5].

Transmission routes of COVID-19:

According to the U.S. Centers for Disease Control and Prevention (CDC) "the principal mode by which people are infected with SARS-CoV-2 (the virus that causes COVID-19) is through exposure to respiratory fluids carrying infectious virus. Exposure occurs in three principal ways [14]:

(1) **inhalation** of very fine respiratory droplets and aerosol particles

(2) **deposition** of respiratory droplets and particles on exposed mucous membranes in the mouth, nose, or eye by direct splashes and sprays

(3) **touching** mucous membranes with hands that have been soiled either directly by virus-containing respiratory fluids or indirectly by touching surfaces with virus on them (fomites)"

Some of the evidence for aerosol transmission of SARS-CoV-2 come from epidemiological studies, and the outbreaks most often cited as indirect evidence of aerosol transmission are the Guangzhou restaurant [25], the Eastern Chinese bus riders [26] and the Diamond Princess cruise ship [27]. A superspreading event at a fitness center in Hong Kong (102 cases) highlights the risk for virus transmission in confined spaces with poor ventilation [28]. The study was unable to identify the predominant transmission mode accounting for the event, but a recent study indicates that physical activities in a fitness center can create high levels of saliva aerosols [29]. These studies are, however, epidemiological case studies and not direct evidence.

Azimi et al. [30] performed a modeling study of SARS-CoV-2 transmission modes on the Diamond Princess cruise ship, utilizing the recently proposed terminology of short-range and long-range airborne transmission routes introduced by Liu et al. [23]. They estimated that the contributions of short-range, long-range, and fomite transmission modes to infected cases were 35%, 35%, and 30%, respectively. Estimates of the contributions of larger respiratory droplets and smaller respiratory aerosols were 41 and 59%, respectively. A conservative assumption of high ventilation rates and no air recirculation conditions for the cruise ship was used in the modeling. In this modeling, the cutoff to distinguish between aerosols and larger droplets was 10 μ m.

Li et al. [31] performed a thorough reconstruction of the Guangzhou incident using both CCTV recordings, *in situ* measurements and numerical simulation to recreate the ventilation flow pattern in the restaurant. All affected guests were seated at three tables in a crowded area that constitutes a separate ventilation zone, in which the airflow was virtually disconnected from the rest of the room. CCTV footage did not record any interactions between the guest at the threes tables, and the distance between the affected persons was sufficiently large to limit the possibility of classical

large droplet transmission. Based on the available evidence, Li et al. [31] concludes that classical airborne transmission of the SARS-CoV-2 virus is possible in crowded spaces that are poorly ventilated.

Other experimental indirect evidence of classical airborne transmission is the identification of SARS-CoV-2 nucleic acids (RNA) in airborne particles smaller than 4 μ m collected in clinical settings [32, 33]. Nevertheless, an obvious prerequisite for aerosol transmission of SARS-CoV-2 is that sufficient amounts of virus-containing aerosols are emitted from the respiratory tract of infected individuals and that, in addition to the viral nucleic acids, the virus remains infective in the airborne state for a sufficiently long period to be inhaled by and establish an infection in others [34, 35]. Controlled animal studies on ferrets showed that transmission of SARS-CoV-2 via both direct contact and air is possible [36, 37]. However, these studies could not discriminate between transmission via small and large droplets.

The exact role and importance of classical airborne transmission of SARS-CoV-2 is still largely unknown, much debated, and a subject to ongoing investigations. Similarly, the role of small droplets for the transmission of influenza virus is also debated. Infectious influenza virus has been detected in small droplets in exhaled breath and coughs and in the air, whereas other studies only recover RNA from air samples and not infectious virus (reviewed in Leung 2021) [38]. Respiratory droplets are further detailed and discussed in Chapter 2.2.

2.1 SARS-CoV-2

Coronaviruses are a group of related RNA viruses that cause diseases in mammals and birds. Most of the viruses cause mild illness such as common cold symptoms [39]. COVID-19 is caused by SARS-CoV-2, a virus first identified in Wuhan, China, in December 2019 [40]. The virus is an enveloped, positive-sense single-stranded RNA Betacoronavirus related to SARS-CoV (causing SARS) and MERS-CoV (causing Middle East Respiratory Syndrome) [25]. SARS-CoV was first identified in China in 2002 and had a case-fatality rate of 7%, whereas MERS-CoV emerged in Saudi Arabia in 2012, and had a case-fatality rate of more than 30% [41]. The individual unit size of a SARS-CoV-2 virion is 60-140 nm [42]. A meta-analysis based on 99 studies estimated that the mean incubation period for SARS-CoV-2 was 6.38 days ranging from 2.33 to 17.60 days [43]. Angiotensin I converting enzyme 2 (ACE2) is a cellular host receptor for SARS-CoV-2 and SARS-CoV in humans. The surface spike glycoprotein (S protein) interact with extracellular domains of the human ACE2 to enter the cells [44]. A study by Ortiz et al. [45] shows that ACE2 receptor is highest within regions of the sinonasal cavity (nose) and in the pulmonary alveoli (lungs), which correlates with sites for virus transmission and severe disease development in the lungs, respectively. The S protein of SARS-CoV-2 is different from the S protein of SARS-CoV and it is assumed to bind more strongly to human ACE2 [46, 47]. The high binding affinity of the S protein surface to the host receptor ACE2 is important for SARS-CoV-2 infectivity [46]. Variants of the SARS-CoV-2 with S protein mutations have emerged [48]. The B.1.1.7 variant, named "Alpha" by WHO [49] (first identified in the United Kingdom), has since spread across the world [50]. Other recently emerged SARS-CoV-2 variants of concern with S protein mutations are the B.1.351 named Beta (first identified in South Africa) and B.1.1.28 named

Gamma (first identified in Brazil). Mutations in the receptor binding domain of the S protein contribute to enhanced binding with the ACE2 receptor, and a specific mutation found in these variants is believed to enhance the transmissibility of SARS-CoV-2 [48]. It is hypothesized that the increased transmission of the B 1.1.7 variant is partly due to the increased binding affinity of the S protein with the ACE2 receptor [51]. Another variants of concern is the B.1.617.2, named Delta (first identified in India).

Current diagnostic testing of SARS-CoV-2 includes detection of virus RNA by PCR assays. Detection of virus RNA by PCR does not provide direct evidence about the presence of infectious virus. Direct evidence for the presence of infectious SARS-CoV-2 can only be provided by the use of cell culture (*in vitro*) assays or an appropriate animal model (*in vivo*) [52]. However, cell culture assays are typically less sensitive and much slower than PCR assays.

2.1.1 Environmental stability

The environmental stability (persistence) of SARS-CoV-2 in respiratory droplets on surfaces and in air can affect transmission of COVID-19. At the beginning of the pandemic, contact transmission via contaminated surfaces (fomites) was particularly feared, and efforts were made to disinfection of surfaces [53]. A publication from Van Doremalen et al. [54] showed that SARS-CoV-2 persisted on surfaces such as plastic, stainless steel, copper and cardboard for various times. The longest persistence was observed on stainless steel and plastic with an estimated median half-life of up to 24 hours. The persistence of SARS-CoV-2 was measured as titer of recovered active (infectious) viruses in cell culture experiments [54], and a high concentrations of virus, 10⁴ infectious virions, was deposited on surfaces. In general, higher temperatures and/or higher relative humidity have a detrimental effect on SARS-CoV and SARS-CoV-2 stability [55, 56], and the decay rate per minutes of viral infectivity was increased from 2% to approximately 4-6% per minute at higher relative humidity or temperatures [56]. At typical air-conditioned environmental conditions (22-25°C and relative humidity of 40-50%) SARS-CoV persisted on smooth surfaces for more than five days. It has been argued that infections through fomites are unlikely in real-life situations. In recent reviews published by Kampf [57] and Goldman [15], it is concluded that the chance of SARS-CoV-2 transmission through surfaces are small. The methodology of many studies on fomites was critically reviewed, stating that the concentrations used were very high in comparison to real-life situations [58]. This conclusion is further supported by studies performed by Mondelli et al. [16-18] whose attempts on using cell culture to propagate virus from swipe-samples collected in hospital settings, where COVID-19 patients were treated, were mostly unsuccessful. Nevertheless, at the same time the authors emphasize the importance of cleaning and disinfection procedures, and the use of proper respiratory protection in the health care setting.

The transmission potential of SARS-CoV-2 via respiratory droplets in air is influenced by the period of time during which SARS-CoV-2 can remain infectious in the airborne state. Laboratory studies suggest that the persistency of SARS-CoV-2 in airborne droplets will be influenced by environmental conditions such as temperature, humidity and sunlight [54, 56, 59, 60]. Under conditions similar to those expected for air-conditioned indoor environments, the decay rates of SARS-CoV-2 were reported to be less than 3% per minute and the time needed for 90% loss of

infectious virus was more than 75 minutes [54, 59]. Smither et al. [59] studied the decay rate of aerosolised SARS-CoV-2 in cell culture media or in artificial saliva at medium (40-60%), and high relative humidity (68-88%). In cell culture medium, SARS-CoV-2 was more stable at medium relative humidity (decay rate of 0.91% per minute) compared to higher relative humidity (decay rate 1.59% per minute). In artificial saliva, however, the decay rate was 2.27% and 0.4% per minute, at medium and high relative humidity, respectively. Dabisch et al. [56] demonstrated that under conditions representative of those expected indoors, or outdoors during nighttime, the decay rate of airborne SARS-CoV-2 was low and the time needed for 90% loss of infectious virus was more than two hours. By contrast, the time needed for 90% loss of infectious virus at high intensity simulated sunlight (40°C, 20% relative humidity) was approximately 5 minutes [56]. Similarly, Schuit et al. [60] also observed that simulated sunlight rapidly inactivated SARS-CoV-2 in aerosols with half-lives of less than 6 minutes, and 90% of the virus was inactivated in less than 20 minutes. An earlier study [61] showed that the half-life for inactivation of human coronavirus 229E decreased from approximately 67 hours at 50% relative humidity to 3 hours at 80% relative humidity. In a more recent study by Schuit et al. [62], it was shown that the aerosol stability of SARS-CoV-2 has not varied greatly among the currently circulating virus lineages, including the B.1.1.7 Alpha mutant.

It is challenging to estimate the impact of environmental conditions on risk associated with surfaces as well as classical airborne transmission of SARS-CoV-2 based on laboratory studies only, since neither the artificially generated virus-laden droplets nor the environmental conditions will be perfect representations of real-world environments or situations.

2.1.2 Infectious dose and viral load

The transmission characteristics of SARS-CoV-2 make it difficult to control the disease. The SARS-CoV-2 appears to be shedded both by asymptomatic and pre-symptomatic infected individuals, and superspreading events are frequently observed [63]. In Hong Kong, several clusters of superspreading events were identified and characterized, and it was estimated that 10-20% of the patients accounted for 80% of all SARS-CoV-2 transmission [64]. Superspreading has also been observed for transmission of SARS and MERS [65]. The number of SARS-CoV-2 virions needed to start an infection, also known as the infectious dose, is still unknown [65]. The diversity (age, underlying health condition etc.) of COVID-19 patients also makes it challenging to converge on the infectious dose that will apply to everyone from observational epidemiological data only [66]. The effect of the new variants, such as the Alpha, Beta, Gamma and Delta on the infectious dose of SARS-CoV-2 is still unclear.

While infectious dose and persistence of SARS-CoV-2 is highly relevant for the back-end of any potential transmission route, another relevant parameter at the front-end is the virus concentration found in the bodily fluids of the host, such as respiratory tract secretions, at different stages of the infection. A meta-analysis by Cevic et al. [67] showed that SARS-CoV-2 viral load appeared to peak in the upper respiratory tract within the first week after symptom onset, and later in the lower respiratory tract. In contrast, the viral load of SARS-CoV peaked at days 10–14 of illness and that of MERS-CoV peaked at 7–10 days of illness. Viral RNA was detected in the upper airways, in the lower airways, in stools and in serum for weeks after infection, however, the infectious virus

was not detected nine days after onset of symptoms in the patients, despite persistently high virus-RNA loads. Note that the diagnostic PCR test detecting virus-RNA does not reflect the presence of infectious virus in a patient, as the presence of RNA can remain long after the disappearance of infectious virus [68]. Recently, Kim et al. [69] isolated in tissue culture the SARS-CoV-2 10 days after onset of symptoms, and in an observation study of a patient, the viral culture remained positive through day 15 and was negative on day 22 from symptom onset [70].

High viral load of SARS-CoV-2 was also reported among asymptomatic and presymptomatic individuals [71]. Notably, asymptomatic individuals with high viral load could potentially be super-emitters of SARS-CoV-2, however, their role in superspreading events is not clear. Oran and Topol [72] reviewed 16 infectious cluster events and concluded that asymptomatic persons probably account for approximately 40% to 45% of SARS-CoV-2 infections. Recent studies show that asymptomatic patients are responsible for fewer secondary infections than estimated early in the pandemic [71]. In an Argentine cruise ship, a febrile passenger emerged, and after testing of all 217 passengers and crew members it turned out that 59% (128) tested positive for COVID-19, of whom 81.3% (104) were asymptomatic [72]. Pan et al. [73] studied respiratory samples from 80 individuals in different stages of the infection, and overall viral load early after onset of the infection was high (more than 10⁶ copies/mL). The viral load peaked around 5-6 days after onset of symptoms. In patients with SARS (SARS-CoV), the viral load normally peaked at around 10 days after onset of symptoms [74]. Avadhanula et al. [75] identified a small group of individuals with mild symptoms and with high viral load (up to 10¹⁰ copies/mL) of SARS-CoV-2. The viral load varied between 10⁴ and 10¹⁰ copies/mL [75]. Wölfel et al. [76] also observed viral concentrations in this range in patient samples.

It is surprising that despite the high viral load in the airways during COVID-19, few published experimental studies have so far succeeded in recovering infectious virus from air samples. The majority of the studies have detected viral RNA in air samples and on no-touch surfaces in health care facilities [34]. Wölfel et al. [76] were unsuccessful in obtaining a viral culture from samples with a viral load below 10⁶ copies/mL. In other studies, no positive culture was obtained in samples with quantitative RT-PCR cycle threshold (Ct) values higher than 24-34 (approximately 10^4 - 10^6 copies/mL) with the fraction of culture positive samples declining with increasing Ct values [67]. For comparison, Milton et al. [77] only detected infectious influenza virus in exhaled breath samples with high (approximately 10^4 - 10^5) copy numbers by quantitative RT-PCR. Technical limitations associated with air sampling and viral tissue culture may partly explain the low success rate of such experimental studies. In particular, it is difficult to preserve the biological state of the collected virus particle. The relatively high limit of detection of cell culture assays may present an additional challenge. As a result of these factors, the potential for transmission of SARS-CoV-2 by inhalation, and particularly aerosol transmission, may be underestimated. Nevertheless, it has recently been possible to detect infectious SARS-CoV-2 from air samples collected close to COVID-19 patients in a hospital environment [78].

2.2 Transmission of SARS-CoV-2 by respiratory droplets

Transmission of viral respiratory diseases is caused by the transport of virus-laden respiratory droplets expelled by an infected individual during respiratory activities such as breathing, talking, singing, coughing, and sneezing. These respiratory droplets consist of liquid fluid, dissolved salt and solid residues, as well as infectious virus particles and viral residues [79]. Droplets of all sizes is subject to evaporation and thus reduction in size. The rate of evaporation depends on droplet size, temperature and the relative humidity of the air [80]. Specifically, the rate of evaporation increases with increasing temperature and with decreasing humidity [80]. The presence of salt and solid residue prevents a complete evaporation. Consequently, the final size of the resulting droplet nuclei is in the range 20-50% of the original diameter [79, 81-84]. The droplets or droplet nuclei may stay suspended in the air or fall to the ground depending on their size and the ambient conditions [80, 85]. In the remaining text, we will refer to small droplets, small particles, small droplet nuclei, and aerosols interhangeably.

2.2.1 Droplet generation

Respiratory droplets are believed to be generated by means of a liquid sheet breakup from the mucosal layers coating the respiratory tract during breathing, talking, and coughing (see Mittal et al. [86] and references therein). Different activities yield different instantaneous concentrations. Duguid [81] reported that a sneeze releases about one million droplets, about 200 times more than for a single cough. Because of the short duration of coughing events, more droplets can be released during speech than during coughing. For example, counting aloud from 1 to 100 is reported to release three times as many droplets as a single cough [87-89].

In the literature, there are large discrepancies in the characterization of droplet sizes between different sources. According to Nicas et al. [82] and Johnson et al. [83], early works such as Duguid [81], Loudon and Roberts [90], and Papineni and Rosenthal [91] suffer from methodological weaknesses. Johnson et al. [83] found that the size distribution of the droplets produced during speaking and coughing has three distinct peaks. For speech, the peak droplet diameters, before evaporation, were found to be 1.6 µm, 2.5 µm, and 145 µm. For coughing, the corresponding peaks were found at 1.6 µm, 1.7 µm, and 123 µm. The first peak relates to the socalled Bronchiolar film burst, a mode also found in regular breathing. The second, referred to as the Laryngeal mode, is most active during speaking and coughing. Finally, the third so-called Oral cavity mode is also active during speaking and coughing. We show the combined tri-modal droplet size distribution function in Figure 2.1. Contrary to the decrease in droplet count in the sub-micrometer range observed by [83], several recent studies (Fabian et al. [92], Lindsley et al. [93], and Asadi et al. [94]) find the highest droplet counts in the smallest size classes (below 0.7 μm). None of these results appear to be corrected for evaporation, which renders comparison of the droplet size distribution from different sources difficult. Neverheless, all these studies find the highest droplet count in the smallest size range measured. This is an important finding, since the review from Fennelly [95] indicates that pathogens predominate in small droplets. Several of the reviewed studies appear to have sampled droplets at a distance, which may make the results susceptile to bias because of fallout of the larger droplets.

Asadi et al. [88] found that the particle count correlates positively with loudness, regardless of language, but that the droplet size distribution is relatively unaffected. Whispering the vocal /a/ resulted in an instantaneous particle concentration of approximately 500 particles per litre whereas shouting gave just over 3000 particles per litre. Between these two extremes, an approximate linear relation with amplitude was seen. Reading an English passage resulted in an average concentration of 200 particles per litre, but there was a large variability between different subjects. They also found that a small fraction of individuals expel a significantly larger number of droplets than the rest, and thus behave as speech superemitters. We refer to Chapter 2.1.2 for a discussion of this topic and for further references.



Figure 2.1 Tri-modal droplet size distribution function according to Johnson et al. [83].

There is clear support in the literature that more than half of the expelled droplets are smaller than 10 μ m [81, 83], and this has a at least two important implications. First, droplets in this size range may hover in the air for very long periods of time and may thus contribute to a background concentration in the surrounding air. Second, droplets smaller than 4 μ m comprise the so-called respirable fraction able to deposit in the bronchiole and alveoli [20].

Lee [96] discussed the minimum droplet size able to contain SARS-CoV-2. He assumed spherical droplets and spherical virions, and used established size data of the virus (70–90 nm [97, 98]) and concentration data from patients (7 x $10^6 - 2.3 \times 10^9$ copies per mL [76]. Assuming a uniform concentration of virus, for the highest viral load the author estimates the minimum droplet size that may contain one SARS-CoV-2 virion would have a diameter of 4.7 µm. Lee's calculation contains an error by a factor of two¹, meaning that the correct estimate based on these assumptions

¹ This has been confirmed by the author (personal communication).

should be 9.3 µm. There are indications that this estimate is too large since viral material has been detected in much smaller droplets. For example, Liu et al. [33] detected viral RNA in a large fraction of droplets smaller than 0.5 μ m in air samples taken from a Wuhan hospital. Lee [96] discusses these results and speculates that they can either be explained by evaporation of the droplets or that the model estimates are based on a too low viral concentration. It is unlikely that evaporation can reduce the size of respiratory droplets by an order of magnitude, since the accepted final droplet size after evaporation is in the range of 20–50% [79, 84]. Using a final size of 20%, the model in [96] would require a hundred-fold increase in the viral concentration, which is not supported in the literature [75, 76]. Lee's analysis does not take into account that the small droplets dominate the respiratory droplet size distribution, and that virions may be enclosed in any droplet that is larger than the virion size, i.e. approximately 100 nm. The minimum size obtained must therefore be interpreted as a statistical measure, and Lee's estimate should therefore be interpreted such that, statistically, we expect to find one virion in each droplet of 9.3 μ m diameter. Likewise, if we consider a smaller droplet of size $2 \mu m$, we expect to find one virion in every hundred droplets. Combining these data with the size distribution of respiratory droplets in Feil! Fant ikke referansekilden., which shows that there are approximately 2000 times more droplets of size 2 μ m than of size 9.3 μ m, we estimate that it is twenty times more likely to find a virion in droplets of size 2 μ m than in droplets of size 9.3 μ m.

2.2.2 Droplet transport

In his seminal work on droplet evaporation and settling, Wells [85] reported that droplets with diameters larger than 100 μ m would settle to the ground in liquid form within seconds. Droplets with diameters smaller than 100 μ m would evaporate into droplet nuclei and stay suspended in air. According to Bourouiba [99], this led to a physics based dichotomy that classified respiratory transmission into a large- versus small-droplet route. In this isolated droplet view, it is easy to realize that larger droplets (larger than 150 μ m) can reach 1-2 m before they impact the ground, whereas smaller droplets (smaller than 100 μ m) encounter significant aerodynamic drag that limits their initial travel distance distance to a few decimeters². This view has led to the conjecture that most droplets are unable to travel further than two meters as a result of respiratory activity.

Recent experiments [99-101] and simulations [102], however, contradict this conjecture. Bourouiba [103] claims that for violent sneezes, the flight path of droplets of all sizes may extend up to 7-8 m, even in the absence of background airflow. Likewise, the droplets expelled by coughing may reach 2-3 m [104, 105]. The reason for this discrepancy is related to the presence of the respiratory jet; the expelled droplets are suspended in a fully turbulent humid and warm puff cloud [99]. This cloud initially has the form of a turbulent jet that will be of different strengths depending on the type of respiratory activity (from breathing to sneezing). As the jet slows down, buoyancy effects will at some point become dominant. This cloud is able to augment the travel distance for droplets of all sizes. Furthermore, the cloud is warm and humid thus delaying the evaporation process. As the jet disperses downstream, the turbulence levels decrease resulting in a continuous fallout of droplets. The smallest droplets may, however, become trapped in the

 $^{^{2}}$ These small droplets will however be suspended in air for long times and may be transported considerable distances by the background flow.

buoyant cloud and stay suspended for a long time. We depict the differences of the traditional isolated droplet view and the modern multiphase view in Figure 2.2.

Abkarian et al. [106] showed that also speech can generate turbulent jets able to reach up to 1-2 m from the host. Although they did not study droplet transport directly, they postulated that small droplets can be transported a distance of 2 m by speech jet puffs during a conversation. This seems reasonable given that droplets in this size range have little inertia. To the best of our knowledge, data are lacking to be able to quantify the maximum droplet size that this type of jet can entrain. Building on the findings in [106], Yang et al. [107] discuss improvements to the standard "wellmixed"³ model for infection probability. In their proposed model, they account for the higher concentration of virions present for people in zones affected by the speech jet. The authors applied their exposure model to a poorly ventilated space for typical droplet production rates and viral saliva loads reported in literature. The exposure model showed that due to the speech jet it takes only 8 min at 1 m distance from an infected subject to reach a high risk of infection. Similarly, the time needed to reach a high risk at a separation distance of 2 m was 16 min. It is not straightforward to compare these results to the risk experienced in well mixed background concentration conditions, but a rough estimate can be found by using the same framework and basic assumptions as Bazant and Bush [108]. For two persons occupying a poorly ventilated room (0.3 ACH) with dimensions 9 m² \times 2.4 m, we find that the time needed for high risk of infection in such well mixed conditions is just short of two hours. Hence, at distances shorter than two meters, the time needed to experience a high risk of infection is at least a factor of ten times shorter when directly exposed to the speech jet than for well mixed, poorly ventilated, conditions. From these results, it is straightforward to conclude that the risk of infection is greatly enlarged when exposed directly to any kind of resperatory jet.

The processes of inhalation and exhalation are fundamentally different. Whereas exhalation produces a strong directed jet, inhalation draws in air almost uniformly from a hemispherical region in front of the head [101], as illustrated in Figure 2.3.



³ In this context, well mixed means that the concentration of virions is uniformly distributed throughout a room.



Figure 2.2 Schematic depiction of the two models of exhaled droplet transport for all respiratory modes. The top frame shows the classical isolated droplet model after Wells [85], whereas the bottom frame shows the more realistic turbulent multiphase model after Bourouiba [99]. Note that different respiratory activities differ in velocities as well as droplet distribution.



Figure 2.3 Conceptual view on the difference between exhalation and inhalation. Exhalation results in the formation of a turbulent respiratory jet with high velocities, whereas inhalation draws air from a hemispherical region in front of the face with lower velocities. Figure is based on Abkarian et al. [101].

2.3 Source control versus respiratory protection

When discussing the benefit and efficacy of different types of face coverings in terms of controlling and preventing transmission of respiratory droplets containing SARS-CoV-2, it is important to distinguish between the two main contexts of their use, namely *source control* and *respiratory protection*.

A COVID-19-infected individual will, depending on the stage of the infection, emit respiratory droplets containing SARS-CoV-2, which upon inhalation by a non-infected individual may pose a transmission risk. In particular, infected individuals who remain asymptomatic or develop only mild symptoms can unwarily infect others [109]. In order to reduce transmission of respiratory

diseases, a common approach involves efforts to stop the emission of these respiratory droplets at its origin – namely the infected host – by using a face covering such as a MFM or CFM to provide *source control*. According to the relevant standards the intended use for MFMs and CFMs is first and foremost to control and prevent the outward emission of respiratory droplets by introducing a physical barrier that provides a filtration mechanism and also dampens the propagation of the respiratory jet.

At the same time, the respiratory tract is a gateway for many air-transmitted factors such as bioaerosols, dust particles, or toxic fumes and gases, which may pose an inhalation hazard and a threat to human life, health and wellbeing. For this reason it is often necessary, especially in an occupational setting, to provide *respiratory protection* for individuals that may be exposed to such inhalation hazards. This is typically achieved using personal protection equipment such as respiratory protective devices (RPDs). Several kinds of RPDs, including filtering facepiece respirators (FFRs), are used to provide respiratory protection by filtering contaminated air, thus stopping e.g. potentially virus-laden respiratory droplets from being inhaled by a wearer. The intended use of RPDs is reflected in their performance requirements and test procedures. At the same time, even though MFMs and CFMs are made of materials with filtering properties, it is neither clearly defined nor understood if, or to what degree, MFMs and CFMs offer any respiratory protection to the wearer. These devices are not classified as RPDs and the performance requirements and test methods applied to these devices according to the relevant standards do therefore not consider this function.

Source control and respiratory protection

According to the U.S. Centers for Disease Control and Prevention (CDC), two conceptual approaches can be distinguished with respect to the use of face coverings as a mean for control and prevention of COVID-19 transmission [110]:

Source control protects others. Source control refers to the use of face coverings (medical face masks, MFMs, community face masks, CFMs) to cover a person's mouth and nose and to help reduce the spread of respiratory droplets to others when the person talks, sneezes, or coughs. This can help reduce the spread of SARS-CoV-2, the virus that causes COVID-19, especially by someone who is infected but does not know it.

Respiratory protection protects the wearer. Respiratory protection refers to respirators (respiratory protection devices, RPDs), which are protective devices that cover a person's nose and mouth or the entire face or head to help reduce the wearer's exposure from breathing in air that contains contaminants, such as respiratory droplets from a person who has COVID-19. This type of protection can include filtering facepiece respirators (FFRs), like N95 respirators.

3 Face coverings against biological contaminants in air

In this chapter, a brief description of different types of face coverings (masks), including their intended use, performance requirements, and testing methodology, is presented with reference to applicable European directives, regulations and harmonized test standards. It should be mentioned that some of the terminology is often used inconsistently in the literature and may even differ between countries and regions. The most consistent terminology can be found in documents issued by relevant national and international standardization bodies. Conformity assessments of commercially available face coverings are generally carried out on the basis of tests performed according to harmonized performance requirements and test procedures defined in standards issued by e.g. the European Committee for Standardization (CEN), the International Organization for Standardization (ISO), and the American Society for Testing and Materials (ASTM). In Norway and in the European Union (EU), RPDs such as FFRs require certification, including an EU Type Examination Certificate by a Notified Body, since FFRs are products falling under the European Regulation for Personal Protection Equipment (EU Regulation 2016/425). MFMs are medical devices that fall under the European Medical Device Directive (EU Directive

93/42/EEC), and therefore do not require third-party certification. Nevertheless, both for FFRs and MFMs, it is ultimately a sole responsibility of the manufacturer to ensure that the product conforms to all the applicable regulatory requirements, and to formally accept responsibility for this by issuing the mandatory EU Declaration of Conformity document.

3.1 Basic terms and definitions

Several parameters and terms are used to describe the properties and protective performance of face coverings such as MFMs, CFMs and FFRs. For this reason, it is useful to introduce some key terminology covering their classification, protective performance and standard testing procedures. A short introduction to FFRs, a specific class of RPDs, is also included, since they are frequently used as a benchmark when considering the performance of MFMs and CFMs. In addition, standard testing procedures for FFRs are often employed to evaluate the protective performance of MFMs and CFMs in the context of respiratory protection.

3.1.1 Filtration efficiency (FE)

Protective performance of face coverings or filter materials is often given in terms of *Filtration* Efficiency (FE) for solid or liquid aerosol particles of a certain size. In general, during FE testing a stream of aerosol-containing air is flowing at a certain speed through the filter material or face covering (mounted in a holder or donned on an artificial head-form) while the upstream and downstream particle concentrations are measured. For RPDs such as FFRs the test aerosol typically consists of sub-micron droplets/particles of sodium chloride (NaCl) or oil (paraffin oil, corn oil) with defined size distributions, whereas for MFM the test aerosol consists of approximately 3 µm droplets/particles containing living Staphylococcus aureus bacteria. The FE is calculated based on the amount of particles that are retained by the filter material or face covering (downstream concentration) relative to the amount of particles in the air stream (upstream concentration). Depending on the test aerosol used, the FE can be expressed as particle filtration efficiency (PFE), bacterial filtration efficiency (BFE) or viral filtration efficiency (VFE). The FE depends on several parameters such as particle size distribution, airflow and the type of filter material [111-114]. The airflow used for testing is given as the amount of air per unit time (e.g. L/min) or as a face velocity (e.g. cm/s). It should be mentioned that for FFRs the FE is measured for the entire device, whereas for MFMs and CFMs only a selected area is tested.

3.1.2 Fit factor (FF)

FE values alone cannot be used to predict the protection level of face coverings, since the openings (gaps) along the perimeter of the mask may be a source of leakage (i.e. contaminated air may penetrate through these gaps during inhalation and/or exhalation, and thus bypass the filter material). In particular, tight-fitting RPDs, such as FFRs, must form a seal between the mask perimeter and the wearer's face to function as intended. With a sufficient seal, a negative pressure is created inside the mask during inhalation, and the air stream must pass through the filtrating layers before reaching the respiratory tract of the wearer. Therefore, the protection level provided by FFRs depend also on several user-related factors, including shape and size of the FFR relative

to the anthropomorphic facial proportions of the wearer. Proper training and stringent donning and doffing procedures are also crucial. To ensure the mask is properly fitted, a quantitative or qualitative evaluation of *Fit Factor (FF)* is performed, during which the leakage around a faceseal is assessed. For qualitative FF measurements, an aerosol of a substance with a pronounced smell (e.g. isoamyl acetate – "banana oil") or taste (e.g. Bitrex) is introduced and the fit is assessed based on whether the wearer can taste or smell the test substance. For quantitative FF measurements, aerosolized droplets/particles are used (e.g. NaCl or corn oil, see also Appendix-**Feil! Fant ikke referansekilden.**8A.2 for further details regarding testing). The test person dons the FFR and enters an enclosure filled with the test aerosol, while the test particle concentration is measured inside and outside of the FFR.

3.1.3 Total Inward Leakage (TIL)

A more accurate estimation of the protective performance of FFRs in terms of leakage is the measurement of *Total Inward Leakage (TIL)*. According to EN149, TIL "consists of three components: faceseal leakage, exhalation valve leakage (if exhalation valve fitted) and filter penetration" [115]. During TIL measurements a trained RPD user dons the FFR, enters an enclosure filled with air containing a test aerosol or gas, and performs a series of movements (exercises), which simulate usual activities performed while wearing RPDs [115, 116]. The test aerosol typically consists of sub-micron droplet nuclei of NaCl with defined size distributions. The test aerosol concentration is measured inside and outside of the FFR during each exercise, and the TIL is calculated as the ratio between these two concentrations (for details see Appendix-**Feil! Fant ikke referansekilden.**8A.2). It should be mentioned that since MFMs and CFMs are loose-fitting devices that are not used for respiratory protection purposes, no TIL requirements exist for these types of face coverings.

3.1.4 Protection factor (PF)

The protective performance of RPDs can be also expressed as *Protection Factor (PF)*, which in a simple approach is the inverse of TIL. The practical implication of PF is easy to understand since if the PF is e.g. 100, then the concentration of a harmful substance in the surrounding air is reduced 100 times while wearing the RPD. However, if the PF is measured under laboratory conditions, it does not necessarily reflect the level of respiratory protection achieved in the working environment. For this purpose, the *Assigned Protection Factor (APF)* is used [117, 118]. Measured under simulated working conditions, the APF is the realistic level of respiratory protection that can be achieved by 95% of the trained wearers, provided that their RPD is properly fitted and well-functioning.

3.1.5 Pressure resistance

Another important property of face coverings is the *pressure resistance*. Various other terms are also used such as differential pressure, pressure drop and breathability, although the latter is somewhat ambiguous since breathability also may refer to the ability of a fabric to allow for moisture transport. For testing purposes, e.g. for FFRs, the device can be donned and sealed on

an artificial head connected to a breathing simulator, and the pressure difference is measured in the vicinity of the mouth opening [115]. A relatively high airflow is used to ensure a labored breathing will not be significantly obstructed by the FFR. For MFMs a part of the device is spread flat on the holder and a defined area is exposed to the airflow while the pressure drop is measured.

3.2 Alternative methods for filtration efficiency testing

The standard FE testing procedures for RPDs are sometimes used for testing MFMs and CFMs, especially for the sub-micron particle size range. However, these procedures exclusively involve droplets/particles of salt, oil or other chemical substances. For this reason, suitability of FFRrelated standards for assessing FE against an aerosol containing microorganisms (bioaerosol) has sometimes been questioned [119]. To determine whether the aerosol type (chemical vs. biological) would introduce any significant change in the performance evaluation of MFMs, several authors have evaluated the FE of MFMs and FFRs using viral (VFE) or bacterial (BFE) aerosols, and compared the results with those obtained using salt particles or oil droplets (PFE). In a study by Rengasamy et al. [120], six N95 FFRs, three surgical N95 FFRs and three models of FDA-cleared MFMs were tested against bacterial, viral and particulate aerosols. The authors discussed VFE, BFE and PFE test methods and concluded that tests for BFE (S. aureus) and VFE (bacteriophage phiX174) performed according to a modified ASTM F2101 are much less demanding than the NIOSH standard testing procedures for N95 FFRs, in terms of airflow velocity and particle size range. For FFRs the NIOSH test conditions are believed to already represent the worst case scenario, with high airflow and charge-neutralized particles of a size close to the most penetrating particle size, and for this reason an additional VFE or BFE evaluation is not necessary. When MFMs were measured using the same NIOSH N95 [121] method with salt particles, their PFE was in the range of 54.74 - 88.4%, while for comparison their VFE and BFE measured according to ASTM tests were over 99% [122]. By contrast, the PFE values obtained during salt particle tests for all N95 FFRs were higher than 98%, and for VFE and BFE higher than 99%. Similar results were obtained by Wen et al. [119] who developed a method for testing VFE using NIOSH-certified N99 FFRs and found that FFRs are equally efficient against 1.2 µm f2 phage aerosols as it is against solid salt particles in a standard NIOSH FFR test. Likewise, Balazy et al. [123] tested two different models of N95 FFRs from two different manufacturers against a 30 nm bacteriophage MS2 aerosol and concluded that testing with solid salt particles can be used to evaluate the FE also against viral aerosol particles of the same size. These results indicate that the use of a "living" bioaerosol is not necessary for assessment of FE and that testing with charge neutral salt particles give a good indication of the FE against other kinds of aerosols, including bacterial and viral ones. If the conservative methods using neutral salt aerosols are indeed sufficient for reliable prediction of the FE of MFMs against bioaerosols, a direct implication could be that the VFE and BFE of MFMs and CFMs can be evaluated using much less resource- and time-consuming PFE tests, and which may allow for a wider or more targeted particle size range to be covered. In order to establish such equivalent procedures it first has to be clearly defined which testing parameters should be used for evaluation of MFM and CFM performance in terms of particle size distribution, airflow velocity, and most importantly, the pass-fail criteria [124].

3.3 The mechanics of face coverings

Wearing any type of face covering will influence the respiratory flow. The main reason for this is that the porous face coverings offer flow resistance that changes the flow pattern close to the face. In particular, the respiratory jet generated by exhalation is obstructed whereas the changes in the flow pattern during inhalation will be smaller but nevertheless important. We show a simplified sketch of the effects of flow resistance in Figure 3.1. The actual porous flow mechanism is quite complicated to fully describe, but on a macroscopic scale it is often modelled as a pressure drop that depends on the internal geometry of the porous elements (see for instance Tamayol et al. [125]). For exhalation, the flow resistance results in an overpressure in the region between the face and the face covering. Likewise, for inhalation, the flow resistance results in a negative pressure between the face and the covering, but numerical simulations by Xi et al. [126] show that, in the absence of leaks, the additional pressure drop moderately alters the flow pattern during inhalation. Therefore, face coverings appear to have a greater impact on exhalation flows.

Conceptual view on pressure generated velocity



a) A pressure pulse leads to an exhalation flow

Flow resistance: Reduction in effective pressure



- *b)* Flow resistance reduces the effective pressure available to generate flow, resulting in a lower velocity. Due to mass conservation, the velocity is the same before and after the region of flow resistance.
- *Figure 3.1* Schematic of the effects of flow resistance. The pressure-driven respiration flow is affected by the extra resistance provided by the face covering.

Another important aspect for the mechanical description is the principle of mass conservation. An immediate effect of this is that there is a direct relationship between flow velocities and the area available for the flow. Additionally, we must take into account leakage caused by gaps between the face covering and the face, which will change the flow resistance distribution, thus also affecting the local flow field. Gaps will typically have less flow resistance than the filter material that leads to escape of air and small droplets. We show sketches depicting these effects in Figure 3.2 and Figure 3.3.



Figure 3.2 Schematic description of the principle of mass conservation, as applied to an example of exhalation. The magnitude of the mask and gap velocities $(u_m \text{ and } u_g)$ depends on the properties of the gap area compared to the rest of the mask and the exit velocity (u_t) .



Figure 3.3 Effect on exhalation flow and droplets from face covering. Similar effects will be seen during inhalation.

The distribution of flow through the filter and gaps is complicated to describe accurately, but Perić and Perić [127] employed the basic principles discussed above to develop a simple model that describes the velocities through the mask and gaps. Assuming that there is leakage through gaps spanning one third of the mask perimeter, the model predicts that gaps as small as 0.15 mm may result in approximately 2% leakage, and that a uniform gap of 1 mm cause more than half the airflow to pass through the gaps. It is important to note that these results stem from a highly simplified model, and in later chapters we will present experimental data that give a better quantification of mask performance.

3.4 Filtering facepiece respirators (FFRs)

FFRs are a class of RPDs and therefore fall under the European Regulation on Personal Protection Equipment (EU Regulation 2016/425). RPDs can filter the contaminated air (e.g. FFRs, powered air-purifying respirators, PAPRs) or supply clean air to the wearer from an external source, e.g. self-contained breathing apparatus (SCBA). RPDs can have numerous designs and technical solutions (e.g. full mask, half mask, hood or helmet; loose or tight fit; with or without powered air supply etc., see Figure 3.4).

In occupational settings, a respiratory protection program is implemented to ensure the workers are sufficiently protected and legal regulations concerning the proper use and intended function of RPDs are followed [118, 128]. During the COVID-19 pandemic, WHO recommends the use of FFRs with the highest protection levels (FFP3, N99 or N100) by HCWs during aerosol generating procedures, otherwise the use of FFR may be affected by their availability and costs [5].



Figure 3.4 Different types of RPDs: elastomeric a) half-mask and b) full-mask, c) FFR folded with earloops and d) head bands; e) mold extruded with exhalation valve, f) horizontally folded with exhalation valve (Photo:FFI).

3.4.1 Product categories

FFRs consist "entirely or substantially of filtering material" and covers mouth and nose of the wearer, creating a seal between the mask's material and the skin [115]. Different models of FFRs are available with or without a valve to reduce the breathing resistance during exhalation (Figure 3.4 e-f). Most FFRs are disposable (single use), although decontamination methods have been applied to allow for emergency reuse of FFRs in times of critical shortage [129]. In general, FFRs consist of several layers of plastic fiber material, such as polypropylene [10]. The in-between layer(s) made of melt-blown or spun-bonded (non-woven) plastic fibers have filtering properties due to their net charge, which allows them to efficiently capture charged particles [10, 113]. This allows for constructing relatively thin filtering materials with low pressure drop and high FE. Studies show that the electrostatic mechanism can be gradually weakened during the mask use by e.g. contact with water, water vapor or solvents [129], or build up of particulates, which are screening the fiber charge and the mechanical filtration mechanisms (Brownian diffusion for smaller particles, and interception and inertial impaction for larger ones) [111-113]. It is acknowledged that for the mechanical filtration media the most penetrating particle size is approximately 0.2-0.3 µm [112], whereas for the electrostatically charged ones (electret), such as those used in FFRs, this size is smaller than 100 nm [111-113, 130, 131] for the airflow velocities usually used in testing. The most penetrating particle size represents a minimum on the penetration curve, with lower penetration of particles with sizes above and below this size.

3.4.2 Performance requirements and test methodology

FFRs should provide protection against different kinds of aerosols, solid and liquid, and it has been shown that FFRs also are highly effective against viral and bacterial bioaerosols, including SARS-CoV-2 [10, 123, 132, 133]. For certification purposes several FFR specifications are tested, the two most relevant for respiratory protection being FE and TIL [114]. Other requirements, such as flammability, compatibility with the skin, pressure drop across the mask, CO₂ accumulation during breathing etc., and their testing procedures are beyond the scope of this report, and can be found elsewhere [115, 118]. Here, we will shortly discuss the standard testing procedures for PFE and TIL found in the European, US and Chinese test standards.

A summary of standard testing procedures and experimental conditions for FE of FFRs is given in Appendix, Table A.1. Briefly, PFE is established by testing against one or more aerosols, usually solid (NaCl) and liquid (paraffin oil, emery oil). The standard testing procedures for PFE are meant to represent the "worst case scenario", with high airflows of 85-95 L/min (EN, GB, NIOSH) and polydispersed aerosols where the particle size distribution is chosen to represent the MPPS for the filter material [120]. The airflow is similar to a wearer's minute volume under strenuous activity [123]. In most European countries, FFR performance is evaluated according to the standard testing procedures described in "EN149: Respiratory protective devices - Filtering half masks to protect against particles - Requirements, testing, marking" [115]. According to EN149 FFRs are divided into three classes depending on their performance: FFP1, 2 and 3. All classes are tested against both solid particle (NaCl) and liquid droplet (oil) aerosols (PFE of FFP1, 2 and 3 should be 80%, 94% and 99%, respectively). In the US, FFRs are divided into several classes, depending on their performance towards solid particles (NaCl) - N95, N99 and N100 (PFE of 95%, 99% and 99.97%, respectively), and liquid droplets oil) (P- and R-series respirators).

For proper assessment of FFR protective performance knowing the leakage is essential (e.g. TIL, FF, PF, APF). Different standard protocols for measurement of TIL for FFRs are given in Appendix, Table A.2. More about the different methods of measurements, and factors and parameters affecting the leakage and fit testing can be found in e.g. [132, 134-139]. A comparison of TIL, PF and APF for EN149-certified FFRs is given in Table 3.1. In the US, according to Occupational Safety and Health Administration (OSHA), FFRs are expected to have APF of at least 10, regardless of their class [117].

Table 3.1Total Inward Leakage, Filtration Efficiency, Protection Factor, and Assigned
Protection Factors requirements for different FFR classes according to EN149
[115, 118].

Class	TIL (maximum)	FE (minimum)	PF	APF
FFP1	22%	80%	4.5	4
FFP2	8%	94%	12.5	10
FFP3	2%	99%	50	20

To summarize, all the standards set high requirements towards the FE of the FFRs, with FFP2, and N95 having similar PFE against solid aerosols. Nevertheless, the PFE is one of many elements affecting the protective performance, and should be considered in conjunction with the parameters referring to leakage of contaminated air into the FFR, such as TIL, PF and APF. The real-life assessment of FFRs must be done by measuring leakage experienced by an individual RPD user.

3.5 Medical face masks (MFMs)

MFMs, often referred to as surgical masks especially in the US, are classified as medical devices and therefore fall under the European Medical Device Directive (EU Directive 93/42/EEC), which is scheduled to be replaced by the European Medical Device Regulation (EU Regulation 2017/745) as of 26 May 2021. MFMs are therefore not RPDs. In Europe, the manufacturer is responsible for issuing the EU Declaration of Conformity document and ensuring that sufficient technical documentation has been established [140, 141]. In the US, the US Food and Drug Administration (FDA) is responsible for clearing MFMs for the US market based on tests performed according to ASTM F-2100 standard [122, 142-144].

The main purpose of MFMs is to to provide source control by controlling and preventing the emission of respiratory droplets from the wearer to the surrounding environment, e.g. protect the patient from respiratory droplets emitted from the respiratory tract of medical personnel, or *vice versa*, if the patient is wearing the MFMs [145]. Certain MFM classes are splash resistant and can provide protection to the wearer from splashes that could occur during surgical and other medical procedures.

3.5.1 Product categories

MFMs consists of several fabric layers usually made of non-woven polyester fibers, where the outer layer (cover web) protects the inner filter layer supported by a shell layer [146]. It may have different forms, as seen in Figure 3.5. The most common types are pleated surgical masks with elastic ear-loops or straps tied at the back of the head.



Figure 3.5 Examples of different models of MFMs: a) molded cone-shaped, b) duck bill, c) pleated with ties and d) ear-loops (drawing by FFI).

3.5.2 Performance requirements and test methodology

Several properties of MFMs are typically tested including their BFE, pressure resistance and splash resistance. Performance testing of MFMs for the European market are done according to harmonized standard EN 14683:2019+AC:2019 issued by CEN [145]. In the US, MFM tests are performed according to ASTM standards. The method for BFE testing of MFMs is similar for all standards. A summary of different European and international standards used for testing of MFMs can be found in Appendix, Table A.3.

According to the CEN standard [145], during BFE testing, air mixed with a bacterial aerosol should flow through the mask material to a 6-stage cascade impactor. The aerosol consists of droplets/particles with mean particle size (MPS) of $3.0 \pm 0.3 \mu m$ containing living *S. aureus* bacteria, and the airflow is 28.3L/min (close to minute volume under rest or light activity). The BFE is evaluated based on the number of colony forming units observed on the agar plates after at least 20 hours of incubation compared to positive control experiments (performed without the mask material). A splash resistance to blood and a pressure drop across the mask are also tested according to ISO 22609:2004. Depending on the BFE and other test results (differential pressure, blood splash resistance, microbial cleanliness), the masks are divided into different classes/cathegories, as shown in Table 3.2. For comparison, MFM classes according to the US standards are given in Table 3.3.

Table 3.2Performance requirements for medical face masks (MFMs) according to European
standard EN14683:2019+AC:2019.

Test	Type I*	Type II	Type IIR
Bacterial filtration efficiency (BFE), [%]	≥95	≥98	≥98
Differential pressure [Pa/m ²]	<40	<40	<40
Splash resistance pressure [kPa]	Not required	Not required	≥ 16,0
Microbial cleanliness [cfu/g]	≤30	≤30	≤30

*According to EN 14683:2019+AC:2019 Type I MFMs "should only be used for patients and other persons to reduce the risk of spread of infections particularly in epidemic or pandemic situations. Type I masks are not intended for use by healthcare professionals in an operating room or in other medical settings with similar requirements".

Table 3.3Performance requirements for medical face masks (MFMs) according to US
standards ASTM F-2100.

Characteristic	Level 1	Level 2	Level 3
	Barrier	Barrier	Barrier
Bacterial Filtration Efficiency (BFE) [%]	≥95	≥98	98
Differential pressure [mm H2O/cm2]	<5.0	< 6.0	< 6.0
Sub-micron particulate filtration efficiency	≥95	≥98	≥98
(PFE) at 0.1 µm [%]			
Resistance to penetration by synthetic blood	≥80	≥120	≥160
[mm Hg]			
Flame spread	Class 1	Class 2	Class 3

It is worth mentioning that MFMs cleared by the FDA for the US market must have PFE tested according to ASTM F2100 with monodispersed aerosol of 0.1 µm latex particles [147]. It is unclear why exactly this particle size was chosen, and the airflow is somehow arbitrary, since according to the standard, air velocity can be in the range of 1-25 m/s. Thus, PFE values may differ depending on which velocity was chosen by the manufacturer, which makes a direct comparison of the PFE performance between different MFMs difficult. Nevertheless, the results from this test may still provide additional information regarding the FE of MFMs against particles in the sub-micron size range, which is not a mandatory test requirement for MFMs approved for the European market.

3.6 Community face masks (CFMs)

CFMs are also frequently referred to as cloth masks, non-medical face masks, and reusable/washable face masks. CFM are neither medical devices nor personal protective equipment, and they vary widely in their design and performance including material and construction (Figure 3.6). Governments across the world have recommended general mandatory use of face coverings for the general public in situations where a sufficient distance cannot be maintained. In order to avoid shortages of MFMs needed by HCWs, the WHO [5] recommends that the general public should wear a non-medical community face mask (CFM). This type of face covering is also common in some parts of the world as a way to protect against air pollution. However, many questions concerning the use and effectiveness of CFMs in controlling and preventing transmission of SARS-CoV-2 have surfaced, especially whether CFMs can be used instead of MFMs. As a result, different guidelines and scientific publications aiming to prove or disprove the efficacy of CFMs have been published [148, 149]. The two most important documents specifying the expected performance of these masks are developed by the WHO [5], and CEN [150]. Further, national standards have also been developed by some European countries including the French Standardization Association (AFNOR) [151] and the Royal Netherlands Standardization Institute (NEN) [152].



Figure 3.6 Examples of different CFM models. (Photo: FFI).

3.6.1 Product categories

The WHO recommends that cloth masks have a three-layer structure where each layer provides a function. It is recommended that the innermost layer is made of a hydrophilic material, and the outermost and middle layers of a hydrophobic material to enhance filtration and/or retain droplets. Factory-made cloth masks should meet the minimum thresholds related to three performance and design parameters: FE, pressure resistance/breathability and shape. For technical specifications WHO refers to a standard by the AFNOR group [151]. The standard recommends that the materials used in CFMs must withstand at least five cleaning and drying cycles. The standard defines minimum performance in terms of FE (≥70%) and breathability (maximum inhalation resistance of 2.4 mbar and maximum exhalation resistance of 3 mbar). In Europe, a non-binding standard (guidelines) for CFMs was issued by CEN in June 2020 (CWA 17553:2020 Workshop Agreement - Community face coverings - Guide to minimum requirements, methods of testing and use) [150]. The standard is rather careful when describing the protective potential of these masks, stating that their role is to "minimizes the projection of user's respiratory droplets saliva, sputum or respiratory secretions when talking, coughing or sneezing. This community face covering may also limit penetration in the user's area of nose and mouth of the respiratory droplets from external origin without claiming the user protection. It also prevents this user's area from any contact with the hands."

In this respect, it seems that the role of CFMs is mainly source control rather than providing respiratory protection to the wearer, although it may also prevent infection via contact with fomites by hindering the subsequent transfer to the face and nose area by touch. This is even more pronounced in the following statement from the standard: "*The community face covering specified in this document do not fall under as a medical device (MD) within the meaning of Directive 93/42/CEE or Regulation EU/2017/745, nor as a personal protective equipment (PPE) with the meaning of Regulation EU/2016/425.*"

CWA 17553:2020 states that CFMs should cover mouth, nose and chin. Interestingly, unlike for MFMs, the dimensions of the mask are also defined with respect to the anthropometric measurements of the face. There are no recommendations regarding user seal check or fit measurements for these devices, as they are loose-fitting by design.

3.6.2 Performance requirements and test methodology

Testing of CFMs can be performed according to the methods described in CWA 17553:2020, although the standard refers to several other standardization documents issued by CEN, including those for MFMs and FFRs, as alternative methods for CFM testing. This offers more leeway to the manufacturers if they have test methods already established, but at the same time makes comparison of CFM performance more complicated. According to CWA 17553:2020 the PFE of a CFM could as an example be measured for mono- or polydispersed aerosol droplets/particles, and the upstream concentration should then be at least 40 particles/droplets per cm³. More details regarding test conditions for CFMs are given in Table 3.4.
CFM Class	Minimum FE	Aerosol type	Aerosol size	Breathing resistance	Face velocity
level 90%:	≥ 90%	solid (sodium chloride NaCl, Talcum powder, Holi powder, dolomite, PSL) or liquid (DEHS paraffin)	3 (± 0,5) μm	Inhalation 2,4 mbar Exhalation 3 mbar	6 (± 1) cm/s
level 70%:	≥70%				

Table 3.4 CFM testing conditions according to CWA 17553:2020.

3.7 Face shields

Another type of face covering, face shields (visors), have also been used by HCWs and the general public to control and prevent the transmission of COVID-19. Especially at the beginning of pandemic, the stand-alone use of face shields was at least to some extent considered as an effective strategy, but this notion has since been abandoned [153, 154]. Face shields may reduce the risk of transocular transmission i.e. inoculation via contact with conjunctival tissue, and the transmission via mucous membranes of the lips and nose [155-158]. Face shields are also relatively easy to use even by unexperienced PPE users, can be readily produced in large quantities, are reusable and can be easily cleaned and disinfected, hence their use is encouraged and commonplace, especially by the general public in some countries [159, 160]. In terms of source control, Verma et al. [160] showed that face shields slow down the stream of exhaled air, during coughing or sneezing. Whereas large droplets may impact on the face shield, smaller droplets are deflected around the edges of the visor. Similarly, in terms of respiratory protection, Lindsley et al. [161] conclude that face shields can substantially reduce exposure to infectious respiratory droplets, while smaller particles flow around the face shield more easily. Face shields are therefore not an adequate substitute for face coverings based on filtration (filter materials) that are capable of stopping the emission (source control) and/or inhalation (respiratory protection) of both large and small infectious droplets/particles, and should therefore be supplemented by other devices such as a MFM or FFR [162].

3.7.1 Product categories

Face shields are classified as PPE (EU Regulation 2016/425). Their intended use is to protect the eyes of the wearer against splashes and ballistic liquid and solid particles. As such, they fall under category of eye protectors [163, 164]. Face shields are typically mounted on a device such as headband, browguard, helmet or hood.

3.7.2 Performance requirements and test methodology

Face shields must cover a rectangular area in the eye-region and provide a certain viewing area. Depending on their specific use, different optical and non-optical properties of face shields are

tested, such as optical radiation filtration, protection against solid particles of different speed, and droplets and splashes of liquid [163-165]. Face shields primarily protect the eyes and the surrounding face area from mechanical (projectiles) or radiation damages. Thus, no requirements are defined regarding their properties in terms of source control or respiratory protection against inhalable respiratory droplets/particles.

4 Filtration efficiency of face masks

Filtration efficiency (FE) is one of the key properties that governs the performance of MFMs and CFMs. It is therefore crucial to describe and systematically discuss the FE of MFMs, CFMs, and different materials used to manufacture face coverings. The amount of respiratory droplets that penetrates through a face covering is determined by the FE. The FE of MFMs and CFMs will be further discussed in the context of source control and respiratory protection. For further evaluation of the efficacy of MFMs and CFMs it is important to answer the following key question:

• How effective are different materials used to manufacture MFMs and CFMs in filtering (removing/retaining) the respiratory droplets and droplet nuclei from the air stream penetrating the face covering?

The FE of a certain material depends on several factors, including the air velocity and particle size distribution. As discussed in Chapter 2, **Feil! Fant ikke referansekilden.** respiratory droplets and droplet nuclei can occur in a wide size range, including sub-micron particles [88, 93]. There is, however, limited information available concerning the exact particle size range that is the most important, relevant and representative for SARS-CoV-2 transmission. For this reason, the contribution of sub-micron droplets cannot be overlooked, and studies where the FE is quantified over the entire respiratory droplet size range for MFMs and CFMs are of special interest. Notably, some authors focus on nanometer particles that correspond to the size of SARS-CoV-2 virions (approximately 100 nm) [114, 119, 123], even though the virions in reality are contained in respiratory droplets of much larger size.

The ability of MFMs and common materials/fabrics used for CFM manufacturing to filter different particle sizes have been investigated in several studies using test methodology based on existing FFR standards as well as custom author-developed test procedures [114, 119, 120, 123, 132, 133, 147, 149, 166-168]. Based on these findings, we will describe and discuss the current state-of-knowledge regarding the FE of MFMs and CFMs.

4.1 Filtration efficiency of medical face masks

The standard test and evaluation methods used to determine the FE of MFMs are derived from and associated with their intended use, i.e. as a mean for source control. The FE of MFMs is therefore evaluated against living *S. aureus* bacteria (BFE) delivered as a test aerosol containing approximately 3 μ m droplets/particles, for which a high FE of at least 95% is required [145]. Little is known about the FE for other particle sizes. The requirements and test methods for MFMs are described in more detail in Chapter 3.5.2.

As mentioned before, FFR standard testing procedures are sometimes used to evaluate the FE of MFMs against sub-micron particles/droplets (PFE). These procedures are quite stringent since high airflows and high concentrations of sub-micron particles are applied. Consequently, when MFMs are tested according to these procedures, their FE for sub-micron particles is usually only 50-90% [114, 120, 147, 169].

In an early study by Tuomi [170], two MFMs from the same manufacturer were taped to an artificial head to eliminate the influence of faceseal leakage, and their FE were measured against polydispersed $0.3 - 10 \,\mu\text{m}$ corn oil droplets. Both MFMs reached 50% FE for 2 μm particles and 95% for particles larger than 5 µm. However, their performance was significantly different for the sub-micron particles - one of the MFM had 30-50% FE for the particles between 0.3 and 2 µm size, whereas FE for the other was almost zero in this size range. Furthermore, the MFM with poor FE for sub-micron efficiency used in the study did not reach the required minimum FE for 3 μ m particles, and had a sharp drop in FE at around 2 μ m. Weber et al. [171] examined eight MFMs from four different manufacturers and found large discrepancies in their performance, with FE between 1-40% for 0.2 µm and more than 98% for 4 µm corn oil droplets. They also noticed an increase in FE at around 2.3 µm droplets for some of the MFMs tested. Similar variations were observed by Chen and Willeke [146], when testing two MFM models with corn oil droplets ranging from 70 nm to 5 µm. Both MFMs had an FE exceeding 90% for droplets larger than 3 μ m, whereas in the sub-micron range the FE was 20% for one mask and 40-80% for the other, depending on the airflow. A plausible explanation could be that for the lower airflows, a higher FE was archieved due to a more efficient capturing of these particles by electrostatic filtration mechanism.

In more recent studies, Oberg and Brosseau [147] tested the FE of six different MFM models, all FDA-cleared with a reported BFE of at least 95% and PFE higher than 97% when measured against approximately 3 μ m *S. aureus*-containing particles and 0.1 μ m latex particles, respectively. They found that the performance of the MFMs was rather variable, with PFE higher than 95% measured with 0.895, 2.0, and 3.1 μ m monodispersed latex particles, and PFE of 60-90% measured with 0.3 μ m salt particles according to the standard NIOSH N-series FFR testing procedure. Similarly, Drewnick et al. [169] tested three MFM models and found their PFE to exceed 80% for 1 μ m salt particles. In the sub-micron range, a large variation was observed in the FE (30-95%), with the poorest performing mask having an FE as low as 30-40% for the 0.3 μ m particles (depending on the air velocity), indicating that the most penetrating particle size was in this range.

A standard NIOSH method with 75 nm NaCl particles was also used by Rengasamy et al. [120, 132] where FEs of 55–90% [120], approximately 98% and under 10% [132] were measured depending on the MFM model. He et al. [114] tested the FE of MFMs exclusively against approximately 20-500 nm particles and found that the FE was 80-90% for airflows ranging between 15-85 L/min.

These results confirm that for MFMs the FE is relatively high (95% or higher) for particles that are 3 μ m or larger, but tends to be lower in the sub-micron range [114, 120, 132, 146, 147, 170, 171]. The observed variation may be caused by the specific filter material used for mask production. In addition, for some MFMs the most penetrating particle size was 45 nm, while for others it was approximately 350 nm, which corrensponds well with electrostatic and mechanical filtration mechanisms, respectively [120, 132]. Hence, it seems that different masks operate under different filtration mechanisms. There is a lack of knowledge about the FE of MFMs against smaller particles and the factors potentially affecting their filtering performance in the sub-micrometer particle size range.

4.2 Filtration efficiency of community face masks

During the COVID-19 pandemic new guidelines for CFM testing and classification have been published (see Chapter 3.6 for further details). It should, however, be pointed out that these guidelines are a recommendation, and not binding requirements. CFMs are often home-made and many of them have been sold with limited information regarding their FE. It is unclear to what degree and how large a share of the CFMs produced and sold on the global market actually conform to the recommendations provided by e.g. CEN or the WHO.

The FE of CFMs depends on the fabrics (fiber characteristics, including diameter, charge, and packing density) in the face coverings. The experimental studies addressing these questions have often not included sufficiently detailed information about the fabrics characteristics (e.g. thread count, type of weave, and thickness), and their relation to the observed FE, which makes direct comparisons of test results challenging. The FE of different fabrics commonly used in CFMs have been reviewed by Clase et al. [172] and Jain et al.[173].

Several studies have evaluated the FE of fabrics commonly used for CFMs manufacturing with particles ranging from 10 nm to 5 μ m [169, 174-178]. The results varied between less than 5% to more than 95% when evaluated across a wide range of particle sizes. The use of multilayer fabrics increased the FE. Drewnick et al. [169] tested the FE of 48 different fabrics, and showed a large variation in FE, from less than 10% to almost 100% for particle sizes ranging from 30 nm to 5 μ m. More than 30 of the fabrics tested had a FE larger than 80% for 5 μ m particles, while only six fabrics had a FE larger than 80% for 30 nm particles. A study by Konda et al. [174] showed that the FE improved when multiple layers of fabrics were used. For specific combinations of different fabrics (hybrid multilayer), such as cotton–silk, cotton–chiffon, cotton–flannel, the FE was larger than 80% for 20–300 nm particles, and 90% for 300 nm–6 μ m particles.

Rengasamy et al. [175] tested different fabrics used in CFMs to establish their FE using 75 nm NaCl aerosols and found that single layers of scarfs, sweatshirts, T-shirts, and towels were associated with a FE of 10-40%.

Wang et al. [179] tested both PFE (75 nm NaCl aerosol) and BFE (approximately $3 \mu m S$. *aureus* aerosol) of different fabrics. PFE of single layer fabrics varied between zero and 23%. When using several layers, the PFE varied between 11 and 56%, depending on the fabrics and their combinations. The BFE varied varied between 16 and 93%, depending on the combinations of fabrics. BFE of single layer fabrics were not tested.

Furuhashi [178] tested the BFE for CFMs consisting of single layers of different types of cottons against an aerosol containing two different bacteria (*S. aureus* and *Serratia marcescens*). The BFE of CFMs ranged from 43-94%, while MFMs that were subjected to the same test had a BFE above 98%.

For CFMs it may be particularly difficult to assess their FE using FFR tests with sub-micron salt particles or oil droplets, because some fabrics can be a source of small particles and fibers, complicating the interpretation of results based on non-specific particle counting and sizing measurements [176].

Filtration efficiency of face masks

Take-aways

- 1. Medical face masks are more than 95% effective in filtering particles larger than 3 μ m, and usually have a filtration efficiency for sub-micron particles between 50% and 90%.
- 2. Community face masks span a wider range of filtration efficiencies, and have been shown to have a filtration efficiency between 15% and 95% for particles larger than 3 μ m, and a filtration efficiency for sub-micron particles between 5% and 90%.
- 3. The filtration efficiency of community face masks depends strongly on the fabrics. The use of multiple fabric layers and combination of different fabrics increase the filtration efficiency for the entire range of particle sizes.

Knowledge gaps

- The filtration efficiency of medical and community face masks for particles in the size range between 1–3 μm is not covered by the standard tests. As a result, there is limited information available on the filtration efficiency in this size range, even though the number of generated respiratory droplets is relatively high in this size range.
- 2. For community face masks, more experimental data are needed to better understand the association between filtration efficiency and different particle sizes, fabric characteristics, and fabric types/combinations.

4.3 Answers to key question in Chapter 4

• How effective are different materials used to manufacture MFMs and CFMs in filtering out the respiratory droplets and droplet nuclei from the air stream penetrating the face covering?

Taken together, the available experimental data show that the filter material in MFMs effectively captures particles that are 3 μ m and larger. This is also to be expected since it is in agreement with the harmonized European test methodology and performance requirements that apply to MFMs (EN 14683).

The FE of MFMs in the sub-micron range is often evaluated using standard FFR test methods. In the sub-micron particle range, there are large discrepancies observed and the reported FE values range from less than 5% to more than 95%. It appears, however, that a majority of MFMs display a sub-micron FE from 50-90%. If we assume a constant FE of 95% for 3 μ m and larger particles and a linear decrease in efficiency towards 50% at 0.01 μ m, we can estimate a total filtration efficiency of approximately 80%.

The FE of fabrics commonly used in CFMs varies greatly depending on both the material properties and the test methods, in particular the particle size used. The FE has been shown to range from less than 5% to more than 90%. The use of multiple layers and/or a combination of different fabrics (hybrid multilayer approach) in CFMs has been shown to increase the FE. CFMs also tend to perform better against particles larger than 3 μ m.

5 Efficacy of face masks as source control against transmission of SARS-CoV-2

To initiate a systematic discussion on how effective MFMs and CFMs are in controlling and preventing the transmission of respiratory droplets containing SARS-CoV-2 when used as a mean for source control, i.e. control/prevent the emission of virus-laden respiratory droplets, it may be useful to break the problem into sub-topics and formulate these as a set of key questions:

- What is the spatial distribution of expelled respiratory droplets with and without face covering?
- What is the size distribution of expelled respiratory droplets with and without face covering for different respiratory activities?
- How much of the exhaled air passes through the filtering material and how much passes through the gaps between the face and the face covering?
- How is the respiratory jet affected by the presence of different face coverings and for different respiratory activities?

In this chapter, we will address these questions in light of the available scientific literature, and describe and discuss the current state-of-knowledge and highlight some knowledge gaps.

5.1 Obstructing the respiratory jet

As discussed in Chapter 2.2.2, the respiratory jet is able to transport droplets relatively far from the host in a short time. For instance, violent sneezes, coughs and speech generate respiratory jets that may extend up to eight meters [103], three meters [104, 105], and two meters [101], respectively. In Chapter 3.3, we also discussed how face coverings are able to block the respiratory jet. Figure 2.2 and Figure 3.3 illustrate these effects. In this chapter, we review experiments and simulations aimed at quantifying this effect.

Verma [180] performed experiments in which they used a manekin head and a manual pump to emulate coughs and sneezes, and a laser sheet illuminated tracer particles seeded at the mouth. Their study provides no quantitative data on velocities or on droplet sizes. However, they show that the penetration length of the cough jet may be reduced from more than 3 m for an uncovered face to less than 25 cm by using any face covering. By seeding tracer particles at the source it is difficult to separate the filtering and the fluid dynamical effects of the face covering. To overcome this weakness, Kähler and Hain [181] instead seeded particles in the test chamber. They show that single un-filtered cough can produce air velocities of about 1 m/s at a distance of 1 m from the source. The use of a face covering reduces the velocities to such an extent that the maximum velocity becomes approximately 0.3 m/s very close to the cougher, and less than 0.1 m/s at distances larger than 0.3 m. They conclude that as long as a face covering has a sufficient flow resistance, it will be efficient in diminishing the respiratory jet.

Tang et al. [182] employed an optical Schlieren method to gain quantitative knowledge of the human cough with and without masks. The advantage of the Schlieren method is that it only relies on thermal gradients, in the air, to capture the flow, bypassing the problem of tracer particles and masks altogether. Their results show that both MFMs and N95 FFRs can efficiently reduce the penetration length of the human cough jet, although the effect of an MFM was redirection and diffusion of the flow as opposed to the blocking effect of the N95 FFR. In both cases, the expelled air rapidly mixed with the rising thermal plume generated by the temperature difference between the surrounding air and the human subject. Using the same type of experimental set up, Viola et al. [105] found that the jet distance from a cough was reduced by about 60-80% for all face coverings without exhalation valves. In yet another Schlieren study, Prasanna Simha and Mohan Rao [104] tracked the front of the cough plume from five volunteers. Although there is large variability between different test persons, both with and without face covering, they showed that there is a universal trend in the jet characteristics. In their experiments, the cough plume could reach up to 3 m from the source and the maximum velocities close to the face was approx. 6 m/s. When the test subjects wore MFMs, the plume distance was reduced to 1.5 meters and the maximum velocities were reduced to approximately 2 m/s. The impact of using N95 FFRs was even larger. In this case, the maximum distance was reduced to less than 25 cm and the corresponding velocity was less than 1 m/s.

Computational fluid dynamics (CFD) simulations is a methodology that has the potential to provide detailed qualitative and quantitative data on the respiratory flow and its interactions with the face covering, as well as for the transport and size distribution of droplets. A major challenge

with CFD modeling, in general, is to make sensible simplifying assumptions to establish a useful computational model, and this is especially true for applications with a high degree of complexity. For example, the porous materials that comprise the mask have a geometric micro-complexity that is difficult to represent accurately with current computational resources and methodology. Current studies [127, 183-186] use empirically determined parametrizations that may have reduced validity for materials with high porosity [125]. Another challenging aspect is that both the face and the face mask are deformable. To the best of our knowledge, methodologies for fluid-structure interactions [187] have not been employed to this problem, and current computational modeling efforts have considered static geometries. At present, it is however not clear what improvements can be achieved by the application of more advanced methodologies.

In the simulations reported by Dbouk and Drikakis [183], the rigid mask has a variable gap along the perimeter and the porosity is modelled as a pressure-loss region with added physics to account for filtering of droplets. They show that the cough jet is effectively redirected by the face mask and that the velocities were reduced to less than one tenth of the unmasked cough velocity. Khosronejad et al. [185] performed simulation of coughing through a face mask. Face masks were modeled as a momentum sink with different pressure losses depending on the mask type. The results indicate that the respiratory jet extends up to approximately 0.5 m, 0.75 m, and 2.5 m for the cases with a MFM, a CFM, and without a face mask, respectively. These results are comparable to some of the experimental results discussed above.

Obstructing the respiratory jet

Take-aways

- 1. Any face covering that provides sufficient flow resistance, i.e. most or all medical and community face masks reduce the forward momentum and redirect part of the flow such that the penetration length of the respiratory jet is reduced by at least 50%.
- 2. The redirection created by medical and community face masks deflect the flow laterally, vertically, and even backwards, causing the exhaled droplets to mix with the ambient air and contribute to the background concentration.
- 3. The combination of fit (faceseal) and flow resistance for a filtering facepiece respirator effectively blocks the respiratory jet and thus reduces the penetration length of the jet by as much as 90%.

Knowledge gaps

- More detailed information, on both the flow field and droplet size distribution, is needed to better understand the role of flow resistance and leakage. This should be possible, even within the framework of current experimental and simulation methodologies, but such data appear to be under-reported.
- 2. Simulation studies could benefit from adopting methodology that take into account the effect of deformability/suppleness of both the face and face mask geometry to assess the uncertainties of simplifying assumptions.

5.2 Mask fit and outward leakage

Since MFMs and CFMs are loose-fitted devices, they are prone to gap leakages. As a consequence, some of the exhaled air and droplets are able to exit through openings around the mask perimeter and thus bypass the filtering layer of the mask. The removal of exhaled droplets are therefore only partly governed by the FE of the mask material that was discussed in the previous chapter.

For a properly fitted FFR without exhalation valve, most of the exhaled air will pass through the filtering layer, removing most of the respiratory droplets. Note, however, that even for a tight-fitting FFR, violent coughs and sneezes may result in an overpressure that leads to intermittent breaches of the faceseal and release of non-filtrated air. On the contrary, for loose-fitting face

coverings, which do not require fitting or fit testing, gap leakages will normally be present, at least in some form, and the size and number of openings will depend on both the face mask and the facial geometries of the wearer. In terms of the flow distribution, the relative importance of the gap leakage increases with increasing pressure resistance of the mask material [127].

Most studies address the source control aspect by measuring the number of emitted droplets that are able to reach the surroundings without distinguishing between FE and gap leakage. There are, however, some studies that have sought to quantify the flow distribution.

Lei et al. used numerical simulations to address the mask fit problem [186] for an FFR, and the fluid flow through the resulting gap leaks [188]. The computational results show that the ratio of mass flow through the filter and through the gaps heavily depend on the breathing velocity as well as the filter resistance. For a pressure drop representative of a N95 FFR, on average 6% of the mass flow passed through the tiny gaps during the exhalation phase of a breathing cycle during normal respiration. Note that this is a substantially higher leakage ratio than the fit requirements of N95 FFRs for inward protection [147]. The authors report that most of the leaks occurred at the nose and at the cheeks. We have not found similar studies that can be used to characterize MFMs and CFMs. Because of the loose-fit design of MFMs and CFMs, higher leakage ratios must be expected.

Similar leakage jets, as in Lei et al. [188], can be seen in the computational cough results from Dbouk and Drikakis [183]. In the same study, they also consider the effect of the face covering on droplet transport. They used the droplet size distribution of Xie et al. [189], with a peak count at about 70 μ m, which means that they have limited support in the small droplet size range. In terms of droplet distance, they find that when wearing a mask, the bulk of the droplets travelled approximately half of the distance compared to not wearing a mask. They do not present results on the distribution of flow and particle flux through the filter and through the gap leaks. Using a similar droplet distribution and mathematical model for the filter, Pendar and Páscoa [184] performed simulations of sneezing in a closed room with two "persons" present. They demonstrated that droplets escaped through any opening in the mask, in particular the larger gaps above the nose. Whereas an unmasked sneeze resulted in droplet transport up to approximately 4 m from the source, the presence of the mask resulted in a spherical contamination region with a radius of approximately 60 cm. However, a major concern about these numerical studies is the lack of validation against experimental data.

Viola et al. [105] used optical Schlieren visualization to experimentally describe the exhaled flow during breathing and coughing for different types of face coverings. They find that MFMs and CFMs generate significant leakage jets. Specifically, their results showed that loose-fitting face coverings could generate intense backward airflow for heavy breathing and coughing conditions.

Bandiera et al. [190] counted ballistic droplets able to hit the ground both for manekins and human volunteers with and without MFMs and CFMs. The size range of the observed droplets was 150–600 μ m. They found that the droplet count for masked test subjects was reduced by a factor of 1000 compared to the unmasked references. Based on this they conjectured that a person standing 2 m from someone coughing without a mask is exposed to over 10 000 times more respiratory

droplets in that size range than a person standing 0.5 m away from someone coughing wearing a single-layer mask. This study shows that even simple face coverings are able to block the transmission of large droplets.

Milton et al. [77] studied source control provided by MFMs against the influenza virus using 37 volunteers. They sampled droplets in the size range $0.05-50 \ \mu m$ in the exhaled air from both masked and unmasked test subjects, and grouped the collected particles into two classes; smaller and larger than 5 µm, respectively. Their results showed that the use of MFMs reduced the viral copy numbers in the large fraction (droplets larger than 5 μ m) by a factor of 25 (higher than 95%) efficiency). For the smaller droplets (smaller than 5 µm), the corresponding reduction factor was 2.8 (approximately 65% efficiency). In other words, MFMs were highly efficient in removing larger droplets from the exhalation flow, but were less effective for smaller droplets. The 65% efficiency observed for the smaller particles may indicate that a substantial fraction of these follow the airflow through the gap leaks, but the FE may also be lower in this size range [147]. We have not found any other studies that address the size-dependent efficiency of MFMs or CFMs that consider a large size range from sub-micron to tens of micrometers. While there is some support in the literature that leakage accounts for 10-40% of inhaled air for MFMs [147, 191], we have not found similar data for exhalation. Due to the overpressure created during exhalation, it is, however, reasonable to assume that the leakage ratio is higher for exhalation than for inhalation.

Kolewe et al. [192] experimentally mimicked exhalation using an artificial head to emit particles. MFMs, CFMs (cotton) and N95 FFRs were tested. Particles in the 0.3-10 µm size range were measured directly in front of, above, and on the sides of the face coverings. Both MFMs and CFMs displayed significant leaks; MFMs primarily sideways and CFMs primarily to the top. As a sidenote, both loosely fitted N95 FFRs, MFMs and CFMs had a leakage of 35-70 times more particles than tightly fitted N95 FFRs. Nevertheless, the study concludes that even poorly fitted face coverings will help reduce the emission of respiratory droplets considerably compared to not wearing a mask.

The studies discussed above, consider only face coverings as source control. A number of studies have considered combined source/receiver experiments on facing manikins in enclosures with varying ventilation rates. Experiments with source/receiver configurations appear to have been first performed by Smaldone and co-workers (Diaz and Smaldone [193], Mansour and Smaldone [194], and Patel et al. [195]), who studied the effect of face coverings in ventilated and unventilated test chambers. They found that for both tidal breathing and coughing, separation between the source and receiver results in significant dilution of the concentration of exhaled droplets. When the distance between the source and receiver was 3 ft (0,9 m) and both were uncovered, the receiver inhaled approximately 1% of the total number of expelled droplets. Note that this dilution appears to be higher than the one reported by Abkarian et al. [106], who, based on a theoretical model of a speech jet, estimated a droplet concentration of approximately 5% at a distance of 1 m. For a ventilated test chamber (approximately 3 liters/s per person), all face coverings used as source control were highly effective [193-195]. Furthermore, MFMs used as source control provided significantly better protection for an unmasked receiver than with an

unmasked source and the receiver wearing a N95 FFR, unless the N95 FFR was perfectly sealed to the manikin [193, 194]. This effect is, however, less pronounced in Patel et al. [195]. The efficacy of face masks as source control for tidal breathing appear to be diminished in the absence of ventilation, possibly caused by build-up of background concentration over time [193]. For coughing, any mask provided significant source control, independent of ventilation.

In another source/receiver study by Ueki et al. [133], two manikin heads were placed 0.5 m apart in a small test chamber (0.24 m³), and one manikin acted as a source of droplets (median size 5.5 µm) containing SARS-CoV-2 virions, whereas the other acted as a receiver. The results showed that when face coverings were mounted on the source, both MFMs and CFMs reduced the viral load experienced by an uncovered receiver by more than 50% and MFMs and CFMs (cotton) performed equally. Covering both the source and receiver did not seem to provide any extra protection to the receiver compared to only covering the source. In a similar study, with a significantly larger, unventilated test chamber (20 m³), Brooks et al. [196] used source/receiver pairs located 6 ft (1.83 m) apart. They used NaCl particles in the $0.1-7 \mu m$ size range. When the source was covered by a MFM, the results showed a 41% reduction in inhaled particles by the receiver. They also considered knotted and tucked MFMs and double masking of the source using a CFM to cover the MFM. As source control, knotted and tucked MFMs showed a 63% reduction in inhaled particles by the receiver, whereas double masking showed a 82% reduction. When both the source and receiver were covered by MFMs, the reduction was 85%, and when both the source and receiver were covered with double masking as described above the reduction was 96%. Although the results in [196] do not explicitly state the amount of gap leakage, one may use the difference between the loose fitted and the knotted/tucked MFMs to provide an estimate of this. If we assume an average filtering efficiency of 80% for droplets over the entire size range, we can infer that gap leaks constitute 20% of the total flow rate for a knotted and tucked MFM and 50% for a regular (loose fitted) MFM.

Among the main differences between the works of Smaldone and co-workers [193-195], Ueki et al. [133], and Brooks et al. [196] are the size of the test chamber, the distance between the source and receiver, and the choice of tracer particles. These can all influence the experimental results and it is unclear to what extent these results can be compared and generalized. For example, the size of the test chamber will influence the background droplet concentration and thus the amount of droplets that can be inhaled by the receiver. Furthermore, the distance between the test subjects will determine whether the receiver will be exposed directly to the respiratory jet puff or to the background concentration. It is also not straightforward to reliably compare the viral titers reported in [133] and the droplet counts reported in [193-196].

Mask fit and outward leakage

Take-aways

- 1. In terms of source control, there is evidence that almost all exhaled large and medium-sized droplets are blocked by medical and community face masks.
- 2. For small droplets there is some evidence that almost half of the exhaled droplets are able to escape to the surroundings.
- 3. Measures to improve the fit of medical and community face masks can improve the source control efficiency by reducing the gap leakages along the perimeter of the mask.

Knowledge gaps

1. Experimental and numerical studies that provide quantitative information on flow velocities and droplets size distributions for different respiratory activities are needed for medical and community face masks. This is important to obtain a better description of the droplet concentrations in the vicinity of an possibly infective individual.

5.3 Answers to key questions in Chapter 5

• What is the spatial distribution of expelled respiratory droplets with and without face covering?

In the *absence of face coverings*, respiratory activities generate turbulent jet plumes in front of the head. Sneezes are the most violent activity, and may lead to the transport of droplets up to 8 m. The jet plume of coughs, and therefore droplets, can reach up to 3 m. It has also been shown that regular speech can generate jets able to reach approximately 2 m. Whereas the latter activity typically occurs for extended periods of time and are very common, the two former are short time events that occur less frequently, especially in asymptomatic individuals. By wearing any type of face covering that offers some flow resistance, the contamination range from these jets are significantly reduced. A conservative estimate from the available literature suggest that for coughs, the jet puff length is reduced by at least 50% (from 2-3 m to 1-1.5 m) for MFMs. Some sources state that the reduction may be as large as 80% (from 2-3 m to 0.4-0.6 m). The plume length itself may be a conservative estimate of the contamination range, since it does not take into account that droplets are filtered by the mask material. Hence, the contamination area may be smaller than the plume length.

• What is the size distribution of expelled respiratory droplets with and without face covering for different respiratory activities?

Respiratory droplets has a tri-modal size distribution with the first two peaks between 1 and 2 μ m, and a third peak at approximately 120 μ m. Experiments show that almost all droplets larger than approximately 5 μ m are captured by the MFMs. Smaller droplets are able to follow the airflow through gap leaks. Additionally, sub-micron droplets may penetrate the filter material. From the available evidence for inhalation, it can be estimated that approximately half of the small size droplets are able to escape during exhalation. However, the presence of the mask effectively reduces the momentum of the flow and the droplets. One is hence left with a situation in which small droplets are ejected with little momentum, and passively follow the airflow of the surroundings. Specifically, they are expected to be captured by the thermal plume rising from the wearer and mix with the surrounding air. A similar assessment can be made for CFMs. There is, however, less knowledge regarding the FE of CFMs over a wide range of droplet sizes, and this increases the uncertainty of the estimates.

• How much of the exhaled air passes through the filtering material and how much passes through the gaps between the face and the face covering?

The amount of flow that passes through the gaps between the face and the face covering is highly dependent on the fit of the mask as well as on the flow rate of the exhaled air. We found one computational study that showed a 6% leakage for FFRs under standard exhalation conditions, but no direct studies on leakage flows for MFMs and CFMs. A rough estimate of the leakage can be deduced from the experiments of Brooks et al. [196], where they compared the performance of loose-fitted and tight-fitted MFMs. By assuming an estimated average FE of approximately 80% for small droplets, which is in line with litterature, we can infer an approximate 50% leakage for the loose-fitted MFM and an approximate 20% leakage for the tight-fitted MFM. We do, however, stress that this is a knowledge gap that needs to be studied using human volunteers and different mask configurations.

• How is the respiratory jet affected by the presence of different face coverings and for different respiratory activities?

As discussed above, face coverings that offer some degree of flow resistance effectively reduces the flow velocity, and therefore obstructs the respiratory jet. Close to the face, MFMs may result in a three-fold reduction of the cough velocity. We have not found solid information on gap leakage velocities, but visualisations have shown that the flow escapes to the side, to the top, and even backwards.

6 Efficacy of face masks as respiratory protection against transmission of SARS-CoV-2

The level of protection offered by MFMs and CFMs against virus-laden respiratory droplets is, at present, not clearly determined [162, 173, 197-202]. While neither MFMs nor CFMs are classified as RPDs, some studies indicate that the use of MFMs may lower the wearer's risk of infection for diseases transmitted via respiratory droplets [198, 203]. In a randomized trial described by Loeb et al. [203], the use of MFMs and N95 FFRs were equally effective in preventing seasonal influenza in over 600 HCWs participating in the study. A recent Cochrane review [198] could not find sufficient experimental evidence to support the need for constant use of RPDs by HCWs during the COVID-19 pandemic. The necessity of using an RPD in a work environment should be based on a proper risk assessment. Furthermore, the Cochrane study seems to imply that MFMs can provide an adequate protection level when used in conjunction with non-respiratory PPE (e.g. face shields, gloves, aprons, etc.) and effective ventilation of indoor areas. Thus, it is difficult to derive any conclusions on the effectiveness of face masks from systematic review studies that do not take into account the underlying physical and biological mechanisms at play.

A key question in order to discuss the efficacy of MFMs and CFMs in terms of their ability to controll and prevent the inhalation of virus-laden respiratory droplets and droplet nuclei, is:

• How much of the *inhaled* air passes through the face covering/filter material and how much passes through gaps between face and face covering?

In this chapter, we address this question in light of the available scientific literature in order to describe the current state-of-knowledge and highlight some knowledge gaps.

6.1 Mask fit and inward leakage

Respiratory droplets suspended in air may penetrate the face covering both through the filter material and through gaps along the mask perimeter. The inhalation flow through such gaps is known as gap leakage. FFRs are designed and tested to minimize gap leakages, whereas loose-fitting MFMs and CFMs are more prone to gap leakage, causing unfiltered air to enter the respiratory tract.

As discussed in Chapter 2, the process of inhalation draws in air and droplets almost uniformly from a hemispherical region in front of the face. In the absence of gap leakage, simulations by Xi et al. [126] show that the pressure drop across the face mask slightly redistributes the inhalation flow pattern, and air enters the mouth and nose through the entire surface of the mask. Leakage at the mask perimeter will change the pressure drop characteristics of the face-face mask system and thus the inhalation flow pattern, but since we have not found any simulation studies that address this issue, we will solely consider experimental studies that have used particle measurements to quantify the performance of face coverings for respiratory protection.

The source-receiver experiments, discussed in Chapter 5.2 in the context of source control, can also be used to assess and compare the efficacy of face coverings in the context of respiratory protection?. The experiments of Smaldone and co-workers (Diaz and Smaldone [193], Mansour and Smaldone [194], and Patel et al. [195]), showed that MFMs when used as source control provided significantly better protection for an unmasked receiver than with an unmasked source and the receiver wearing an N95 FFR, unless the FFR was perfectly sealed to the manikin[193, 194]. In one of the studies, however, this effect was less pronounced [195]. Similar conclusions can be drawn from the study by Ueki et al. [133] who also used two manikin heads in a source-receiver setup. The difference in performance when used as source control and respiratory protection may be related to the asymmetry between the inhalation and exhalation flow patterns discussed in Chapter 2.2.2, and demonstrates that leakage appears to be more important in determining the efficacy of MFMs and CFMs as a mean for respiratory protection than for source control.

A quantitative approach, normally used for RPDs, can be employed to evaluate the faceseal leakage of MFMs and CFMs. The leakage is usually expressed in terms of TIL (total inward leakage), FF (fit factor), or PF (protection factor). Methods for measuring the leakage of RPDs are discussed in Chapter 3.1. There are no formal requirements regarding mask fit for MFMs and CFMs, but, with standard test procedures, their respiratory protection properties can be directly compared to the performance of FFRs. TIL measurements can be performed using either a panel of test persons or artificial heads. These artificial heads usually represent generalized anthropometric features but may impose an additional challenge in fitting the mask, as they are often manufactured in hard non-deformable polymers. Also for human test subject, there are a number of factors that affect the results of the tests: Breathing pattern, variability of facial dimensions, facial hair, gender, ethnicity, and user experience with face coverings [114, 139, 147, 204-209]. For instance, in a study by Oberg and Brosseau [147] where the FF of MFMs was measured for 20 human subjects without previous experience with MFM use, there was a statistically significant difference in the results depending on whether or not the test subjects were given assistance in donning the mask (head strap adjustments, nosepiece tightening), with an average FF of 4.4 and 5.7 for unassisted and assisted, respectively. Furthermore, the study concluded that even MFMs with a high FE had an FF less than 10 in all cases. These FF results are too low for MFMs to be used as a replacement for FFRs, for which the requirement is a FF larger than 100.

Lee et al. [134, 204] performed similar studies on different MFM models for several male and female volunteers. The authors measured PF using 93 nm to 1.61 μ m NaCl particles [204] and 40 nm to 1.3 μ m NaCl particles [134] and observed a large variation in PF results depending on particle size and MFM manufacturer. The PF was less than 10 in the entire size range tested and reached a minimum within the 40 nm to 0.3 μ m particle size range. The size at which the minimum PF occurred, coincides with the size at which minimum FE was observed for MFMs. This indicates that the FE has a direct impact on the observed inward leakage results. Preferably the leakage tests should be performed for particle sizes where the FE is high so that the contribution of the gap leakage would be distinguishable from FE. It seems that for a majority of MFMs this particle size is likely in the range 3 μ m and larger.

The effect of the particle size chosen for TIL measurement was also shown by Rengasamy et al. [132] on a manikin-based study, where TIL for two MFM models and two FFR models were compared. TIL measurements were performed for different sub-micron NaCl particles (20-800 nm) and airflows (8 and 40 L/min) delivered by a breathing simulator. MFMs and FFRs were either sealed on the manikin head (no leakage) or given different-sized artificial leakages to imitate an imperfect fit. One of the MFMs had a very high TIL in the absence of leakage of approximately 45-68% (depending on the flow rate) for 45 nm particles and 88-97% for 300 nm particles. This indicates a poor FE of the mask. In this case, there was virtually no impact on the TIL when adding an artificial leakage. The other MFM had a TIL in the absence of leakage of approximately 4-9% at 45 nm and 2-3% at 300 nm, indicating a high FE. In this case, once leaks were introduced, the TIL was mainly affected by particles entering via openings in the faceseal, resulting in a TIL of up to 37%. All particles in the studied size range were able to follow the flow through the faceseal leaks The observed results of TIL show a small dependence on particle size, which means that size-dependent FE may also play a role.

Grinshpun et al. [205] performed experiments with 25 human volunteers, donning both MFMs and N95 FFRs, during various exercises. Three measurements were performed for each subject. Additionally, the respiratory patterns were recorded and the measurements were replicated in a laboratory set-up using manikin heads. In total, the study comprised 5250 data points. For particle size range up to 1 μ m, the number of particles entering via the faceseal leakage for MFMs was about five times the number of particles passing through the filter. Since this study explicitly state a FE (above 90%), one may readily estimate that this leads to a FF of 2.2, which corresponds to a leakage ratio of approximately 40%. Note that the amount of leakage varied only weakly with particle size, whereas this dependency was well pronounced for the N95 FFRs.

Large openings along the mask perimeter will be a major source of particles entering the face covering [132, 204]. Most of the available studies have focused on tight-fitting RPDs, such as FFRs, and the leaks in the faceseal have been simulated using slits, tubes or other devices to create a relatively small (milimeters or smaller) gap with a controllable size, shape and area [132, 210, 211]. In reality, the shape, size and the site of these gaps would depend on the individual facial features of the wearer and the mask form. To some extent, the variations among facial features can be reflected by using statistical data over a large number of individuals, such as the bivariate panel used by NIOSH based on the face breadth (bizygomatic) and length (Menton-Sellion) for several industrial workers in US [212]. These studies could be used for prediction of the perimeter leaks for larger groups of users (e.g. general public, HCWs etc). Nevertheless, some studies show that this approach may be somehow oversimplified [208, 209]. Therefore, it is unclear what impact the facial dimensions would have on the leakage for loose-fitting devices such as MFMs and CFMs.

Sickberg-Bennett et al. [213] tested a range of MFMs with elastic ear loops and ties by measuring a "Fitted Filtration Efficiency" (FFE)⁴ on volunteers. Different models of MFMs and FFRs were tested for comparison. Volunteers (one male and one female) entered one by one a chamber filled

⁴ FFE is a term not used outside this and other publications by the authors, and is in principle an equivalent of a TIL measurement with results given in terms of filtration efficiency rather than leakage.

with polydispersed 0.5 µm salt particles. The masks with ties provided much better protection than the ones with elastic ear loops (FFE of approximately 70% against less than 40%, or correspondingly a TIL of 30% versus 60%, respectively). A plausible explanation of this is that the masks with ties can be readily adjusted to the individual user's face, probably achieving better tightness and thus indicating the importance of the mask fit. In another study by the same group, Clapp et al. [148] conducted FFE measurements of CFMs, MFMs and FFRs under the same experimental conditions on one adult male volunteer. Several modifications such as adjusting the ear loops by tying them or fastening with a clip, increasing mask tightness by rubber bands or nylon hosiery, or inserting a metal nose bridge were introduced to investigate whether they improve the mask fit. The FFE of different CFMs tested on one adult male with no beard ranged from 80% to 25%, with the modified two-layer nylon CFM having the best results. The FFE of the unwashed nylon CFM of 45% was improved by adding a metal nose bridge to 55% and further to 75% when a filter insert was mounted. On the contrary, the unmodified MFM with elastic earloops yielded a FFE of approximately 40%, which could be improved to approximately 80% both when a nylon hosiery was worn over the mask, and when rubber bands were used for securing the mask. Even better results with improving the MFM fit was achieved by Runde et al. [214], who demonstrated that MFMs secured with three connected rubber bands could pass the N95 FFR fit test. A key take-away from these works is the importance of sealing any leaks in the face covering, which is especially true for loose-fitting devices such as MFMs and CFMs, if they are to provide sufficient respiratory protection. Another important observation is that the studies using human volunteers suffer from a small number of the test subjects with limited variations in their body composition and facial features relevant for fit goodness. For this reason, the results of many studies may not necessarily be representative for larger groups of society.

Mask fit and inward leakage

Take-aways

- Medical and community face masks appear to be less effective for respiratory protection than for source control. This may, in part, be caused by the loose fit of these devices that enables leakage through the gaps along the mask perimeter. The size and shape of these gaps, depend on the facial characteristics of the user.
- 2. For medical and community face masks there are no formal requirements to determine the faceseal leakage.Standard testing procedures for filtering facepiece respirators are not necessarily suitable for community and medical facemask evaluation, since submicron particles are used and there is large variability in the filtration efficiency in this range for different face mask materials.
- 3. The reported fit factor for medical face masks is in the range 2-10, corresponding to a total inward leakage in the range 10-50%. These fit factor results are too low for medical face masks to be used as a replacement for filtering facepiece respirators, for which the requirement is a fit factor larger than 100.
- 4. Reduction of leakage through gaps in the faceseal can be achieved by simple means, such as knotting/tucking and mask braces, and leads to significant improvement in the respiratory protection offered by medical and community face masks.

Knowledge gaps

- 1. Studies assessing the total inward leakage and filtration efficiency independently, spanning a large particle size range, are scarce for medical and community face masks. Such studies would enable the quantification of gap leakages.
- 2. Standard testing procedures for respiratory protection devices are not suitable for the assessment of inward protection for medical and community face masks. Tailored testing procedures for these devices should therefore be developed.
- 3. Studies that quantitatively assess the leakage of medical and community face masks used by a larger and representative group of users should be undertaken. Such studies should include the effect of measures to improve the mask fit.

6.2 Answer to key questions in Chapter 6

• How much of the *inhaled* air passes through the face covering/filter material and how much passes through gaps between face and face covering?

The available evidence suggests that 10-40% of the inhaled air passes through the gap leaks for MFMs. This range is based on fit testing both on human subjects and manikin heads that have reported FF values in the range 2–10, which correspond to a TIL of about 10–50%.

7 The effects of long term use, humidity, washing and reuse of face masks

To initiate a systematic discussion on the effects of long term use and humidity on CFMs and MFMs, and washing and reuse of CFMs, it may be useful to break the problem into sub-topics and formulate these as a set of key questions:

- How long can CFMs and MFMs be used before they get too humid or wet to wear? What is the definition of humid and wet face coverings?
- What is the effect of humidity on FE and fit of CFMs and MFMs?

• Do washing and drying cycles affect the FE and fit of CFMs? How many cycles can be performed?

In the following chapter, we will address these questions in light of the available scientific literature in order to describe the current state-of-knowledge, and to highlight some knowledge gaps.

It is difficult to generalize the effect of long term use on FE and fit of CFMs and MFMs since many of the fundamental properties will depend on the choice of materials (fabrics), the number of layers used, the combinations of these, and the general design [12, 173, 175]. CFMs have different designs, some of them very close to MFMs, whereas others are more similar to FFRs (e.g. duck-bill, horizontally folded). Nevertheless, no performance or evaluation requirements regarding the faceseal are given in the recommended testing procedures for CFMs or harmonized test standards for MFMs [150-152], and they are therefore both considered loose-fitting devices where gap leakages have to be expected.

The efficienty of MFMs and CFMs to remove droplets from the exhaled and inhaled air, depends both on the FE and gap leakage (Chapter 4-6). CFMs and MFMs become humid or wet from the human breath during use, and there is a limited number of studies that have addressed the effect of humidity on the FE of face coverings. In a recent study by Zangmeister et al. [215], the FE of different types of cotton were tested at low (55%) and high (99%) relative humidity. Under humid conditions (99%), the tested cotton fabrics increased their FE from 12% to 45%, with an average increase of 33% when challenged with a 300 nm NaCl aerosol. Synthetic fabrics, such as nylon, polyester and rayon performed poorly relative to cotton and their performance did not improve with increased humidity. The FE of MFMs and FFRs, which typically consist of plastic fiber materials such as melt-blown/non-woven polypropylene, was the same under both low and high humidity conditions. The increased FE of the 100% cotton fabrics can be explained by increased size of the particles due to H₂O uptake and the following increased capture of the particles.

The re-use of CFMs contaminated with SARS-2-CoV can potentially increase the risk for subsequent spreading of COVID-19. For this reason, a thorough understanding of the environmental stability and persistence of SARS-CoV-2 on contaminated masks may have impact on the handling of both reusable and single-use items during use and after wear. Kasloff et al. [216] studied experimentally the stability of SARS-CoV-2 on inoculated surfaces of face coverings used by healthcare workers and members of the general public. Of all the materials tested, cotton provided the lowest environmental stability to SARS-CoV-2. After one hour of drying, the inoculated infectious virus was disappeared almost completely from the input inoculum. However, infectious SARS-CoV-2 was recovered after 21 days on plastic, 14 days on stainless steel, 7 days on nitrile gloves and 4 days on chemical resistant gloves. In addition, significant quantities of infectious SARS-CoV-2 could be recovered from inoculated N95 FFRs for 14 days. Note that a a high titre inoculum (10⁶-10⁷) was used in the study to represent a worst-case scenario of SARS-CoV-2 persistence on a contaminated surface. However, it has been argued that the use of high concentrations of SARS-CoV-2 inoculated on surfaces does not reflect the real situations (Chapter 2.1.1) [15].

The FE and the shape/fit of a CFM must be retained during the washing and drying cycles. The guidance from CDC (2020)⁵ for cleaning says that the CFMs should be cleaned at 60 °C with regular household laundry after each use. WHO refeers to the standard by AFNOR [151] and recommends that the materials used in CFMs must withstand at least five cleaning and drying cycles. Reusability without compromising the FE is crucial, but few studies have examined the influence of cleaning cycles on the FE of CFMs. In a study by Hao [217], eight fabrics were analyzed and the FE before and after 1, 5, and 10 cycles of washing and drying was examined. No measurable change in the structure of fabric materials and no impact on the FE of the fabrics was observed, unlike the FE of MFMs and respirator materials, which degraded drastically already after one cleaning cycle. This decrease may be due to the loss of charges attached on the fibers of the MFMs and respirator materials. However, it was observed that the FE of N95 FFRs and MFMs materials were still higher than many fabric materials used in CFMs (depending on the fabrics). A study by Clapp et al. [148] showed that washing of a two-layer nylon CFM slightly increased the FE from 74 to 79%, whereas Neupane et al. [218] observed that increasing the number of washing and drying cycles gradually decreased the FE of CFMs with a reduction in FE by 20% after the 4th washing cycle. Microscopic examination revealed that the pore size was increased and the shape of the pores were different after the washing and drying cycles.

The effects of long term use, humidity, washing and reuse of face masks

Take-aways

- Increased humidity seems to increase the filtration efficiency of community face masks made of cotton. The filtration efficiency of synthetic fabrics, such as nylon, polyester and rayon did not improve with increased humidity. Increased humidity had no effect on filtration efficciency of medical face masks.
- 2. The effect of washing and drying cycles on the filtration efficiency of community face masks is contradictory. Studies have shown both increased and decreased filtration following washing and drying.

⁵ How to Wash a Cloth Face Covering | CDC

Knowledge gaps

- 1. Experimental studies have not provided enough knowledge about the effect of humidity on filtration efficiency of community and medical face masks.
- 2. More knowledge is needed to estimate how long community and medical face masks can be used as face coverings before they become too humid or wet to be efficient (filtration efficiency and breathability).
- 3. There is a need for more studies investigating the effect of washing and drying cycles on the filtration efficieny of different fabrics, and the risk of increased faceseal leakage (fit) of community face masks

7.1 Answers to key questions in Chapter 7

• How long can CFMs or MFMs be used before they get too humid or wet? What is the definition of a humid or wet face covering?

It is difficult to generalize if, and if so, how long it takes before face coverings become too wet or humid to be efficient. The face coverings will become humid from the human breath, but environmental conditions (weather conditions such as temperature, rain, sunshine) may also have an impact on humidity and efficiency of the face coverings. There is a lack of knowledge concerning how long CFMs and MFMs can be used before they become too humid or wet to be efficient, e.g. due to changes in filtration efficiency and/or breathability.

• What is the effect of humidity on filtration efficiency and fit (increased faceseal leakage) of CFMs and MFMs?

Few studies have investigated the effect of humidity on the filtration efficiency and fit of CFMs and MFMs, and a broad generalization is therefore difficult. It is indicated that increasing the humidity from 55% to 99% increase the filtration efficiency of cotton fabrics, while no effect of humidity was seen for synthetic fabrics or MFMs (typically composed of synthetic fabrics). A possible explanation was increased droplet size due to H_2O uptake inside the fabric, followed by an increase in particle capture.

• Does washing and drying cycles affect the filtration efficiency and fit of CFMs? How many cycles can be performed?

Few studies have investigated the effect of washing and drying cycles on the filtration efficiency of different fabrics used in CFMs, and the effects of washing and drying cycles

reported have been contradictory since studies have shown both increased and decreased filtration efficiency following washing and drying.

8 Discussion and conclusions

The benefit of using different types of face coverings to control and prevent the transmission of SARS-CoV-2 have been debated since the beginning of the COVID-19 pandemic. This study has reviewed the available scientific literature concerning the efficacy of medical fask masks (MFMs) and community face masks (CFMs), with a particular focus on experimental and numerical studies and results.

The basic idea behind using a face covering is to provide a filtering layer between the respiratory tract and the surroundings in order to prevent the inhalation and/or exhalation of pathogencontaining respiratory droplets while still allowing air to pass. The ability of face coverings to prevent the transmission of pathogens is closely connected to the underlying transmission mechanisms. Historically, and throughout most the COVID-19 pandemic, the main transmission guidelines for viral respiratory tract infections have referred to a contact route, a droplet route, and an aerosol (airborne) route. Recent updates to the main transmission guidelines have however redefined the transmission routes as inhalation of virus, deposition of virus on exposed mucous membranes, and touching mucous membranes with soiled hands contaminated with virus.

Respiratory activities such as breathing, talking, singing, coughing and sneezing expel droplets that span a vast size range from sub-micron to about 1 mm. In terms of droplet count, more than half of the droplets expelled are smaller than 10 µm. However, since the volume grows as the cube of the diameter, most of the expelled liquid volume is found in droplets larger than 100 µm. Droplets of all sizes can be ejected at high momentum during coughs and sneezes, and therefore deposit on a receiver, but only large and medium-sized droplets have sufficient mass to fall to the ground and contaminate surfaces. Likewise, only small and medium-sized droplets are light enough to stay suspended and be transported by the air currents. Both small and medium-sized droplets are inhalable, but only small droplets may penetrate into the lower airways. The specific cut-off between large, medium, and small droplets depends on several factors such as air currents, humidity, temperature, and salt/solid content of droplets, and is therefore difficult to assess. In this review, we have used the following classification: Small droplets are smaller than approximately 15 μ m, whereas large droplets are larger than approximately 100 μ m, with a size class for medium-sized droplets in-between. Since the importance of different droplet sizes on transmission of SARS-CoV-2 by inhalation is not well understood, it may be necessary to consider the efficacy of face masks in all three size categories.

Medical face masks are originally intended solely for use as source control, and typically in a clinical environment. During the COVID-19 pandemic, the use of medical and community face

masks have been advocated as source control, i.e. to control and prevent respiratory droplet emission from a wearer, and to some extent also respiratory protection, i.e. to protect the wearer from inhaling respiratory droplets an droplet nuclei. There is, however, a fundamental difference in face covering use for source control and respiratory protection. For source control the face covering is used to minimize the outward emission, with the most important features being to block the large and medium-sized droplets and to obstruct the respiratory jet. For respiratory protection, the face covering is used to protect against an airborne contamination that will be variable and context dependent both in terms of droplet and droplet nuclei size range and concentration level, which means that it is difficult to discern the practical/actual level of protection needed. Therefore, when testing face masks for respiratory protection, it is necessary to consider a large range of droplet sizes, including sub-micron droplets. In practice, for loosefitting medical and community face masks, it is difficult to avoid gap leakages. As a consequence, about half of the air, and the particles contained within it, may be able to bypass the filter material.

In terms of source control, experimental studies have shown that both medical and community face masks can be effective in controlling and preventing the emission of medium-sized to large respiratory droplets. Recently, it has been shown that all exhalation events, such as breathing, talking, singing, coughing and sneezing produce warm, moist droplet clouds entrained in a turbulent jet that can propagate several meters. Sneezes may lead to the transport of droplets up to eight meters, whereas coughs can transport droplets up to three meters. It has also been shown that even regular speech can generate jets able to reach approximately two meters. There is evidence that the use of face coverings reduces the penetration length of these jets by more than fifty percent. This implies that the exhaled air will act as a low-momentum source of primarily small droplets. These droplets will, most likely, rapidly mix with the background concentration. This is in contrast to what happens when not wearing a face covering, where instead a strong directed jet flow is formed that contains a cloud of droplets of all sizes. Note that to achieve complete efficiency as source control, it is important both that the filtering efficiency is high enough for all particle sizes and that gap leakages are avoided. This is usually not achieved for loose-fitting face coverings such as medical and community face masks. In practice, gap leakage leads to the escape of about half of the small respiratory droplets. In addition to the obstruction of the respiratory jet, there are two external effects that contribute to the overall efficacy of face coverings: i) the ability of the ventilation system to reduce the background concentration in indoor environments, and ii) the time the virus remains infective in suspended droplets. Both these effects impact the transmission to individuals in the vicinity.

For respiratory protection, the filtration efficiency across the range of droplet sizes as well as the gap leakage of medical and community face masks is more important than for source control. For medical face masks, the filtering efficiency for droplets that are 3 μ m or larger appears to be high (above ninety five percent). This is not surprising, since it is in line with the minimum performance requirements according to the applicable harmonized test standards. On the other hand, there is no guarantee that this performance will apply also to droplets that are smaller than 3 μ m. The filtration efficiency for sub-micron droplets usually range from fifty percent to more than ninety percent, but some studies have shown filtration efficiencies close to zero percent. For community face masks, the filtration efficiency usually range from fifteen to more than ninty-five

percent for droplets that are 3 μ m or larger, and from five to more than ninty percent for submicron droplets. There are no formal performance requirements for community face masks, but recent guidelines have proposed that such products should be capable of filtering at least seventy precent of particles that are 3 μ m or larger. Gap leakages constitute a major concern for medical and community face masks, since both have a loose-fitted design. Gap leakages as large as forty percent have been reported. Most or all small, and possibly some medium-sized, droplets and droplet nuclei will be able to follow the inward airflow through the gap leakages. This may result in a significant inhalation exposure. It has been shown that simple measures, such as knotting/tucking or mask braces, can lead to a signicant gap leakage reduction for medical face masks.

References

- 1. Cucinotta, D. and M. Vanelli, *WHO Declares COVID-19 a Pandemic*. Acta Biomed, 2020. **91**(1): p. 157-160.
- 2. World Health Organization (WHO). *Coronavirus (COVID-19) Dashboard*. 2021; Available from: <u>https://covid19.who.int/</u>.
- 3. European Center for Disease Prevention and Control (ECDC). Using face masks in the community: first update. Effectiveness in reducing transmission of COVID-19. 2021; Available from: <u>https://www.ecdc.europa.eu/sites/default/files/documents/covid-19-face-masks-community-first-update.pdf</u>.
- 4. US Centers for Disease Control and Prevention (CDC). Use Masks to Slow the Spread of COVID-19. 2021; Available from: <u>https://www.cdc.gov/coronavirus/2019-ncov/prevent-getting-sick/diy-cloth-face-coverings.html</u>.
- 5. World Health Organization (WHO) *Mask use in the context of COVID-19 Interim guidance 1 December 2020.* 2020.
- 6. Butler, D., J. Eilperin, and T. Hamburger, '*No offense, but is this a joke?*' *Inside the underground market for face masks.*, in *Washington Post.* 2020.
- 7. Organisation for Economic Co-operation and Development (OECD) *The face mask global value chain in the COVID-19 outbreak: Evidence and policy lessons.* 2020.
- 8. Patrício Silva, A.L., et al., *Increased plastic pollution due to COVID-19 pandemic: Challenges and recommendations*. Chemical Engineering Journal, 2021. **405**: p. 126683.
- 9. Fadare, O.O. and E.D. Okoffo, *Covid-19 face masks: A potential source of microplastic fibers in the environment.* Science of The Total Environment, 2020. **737**: p. 140279.
- 10. Fisher, E.M. and R.E. Shaffer, *Considerations for recommending extended use and limited reuse of filtering facepiece respirators in health care settings.* J Occup Environ Hyg, 2014. **11**(8): p. D115-28.
- 11. Chua, M.H., et al., *Face Masks in the New COVID-19 Normal: Materials, Testing, and Perspectives.* Research, 2020. **2020**: p. 7286735.
- 12. Lee, K.P., et al., *Reusable Face Masks as Alternative for Disposable Medical Masks: Factors that Affect their Wear-Comfort.* Int J Environ Res Public Health, 2020. **17**(18).
- 13. Morawska, L. and D.K. Milton, *It Is Time to Address Airborne Transmission of Coronavirus Disease 2019 (COVID-19).* Clin Infect Dis, 2020. **71**(9): p. 2311-2313.
- 14. US Centers for Disease Control and Prevention (CDC). *Scientific Brief: SARS-CoV-2 Transmission*. 2020; Available from: <u>https://www.cdc.gov/coronavirus/2019-</u> <u>ncov/more/scientific-brief-sars-cov-2.html</u>.

- 15. Goldman, E., *Exaggerated risk of transmission of COVID-19 by fomites*. Lancet Infect Dis, 2020. **20**(8): p. 892-893.
- 16. Colaneri, M., et al., *Severe acute respiratory syndrome coronavirus 2 RNA contamination of inanimate surfaces and virus viability in a health care emergency unit.* Clinical Microbiology and Infection, 2020. **26**(8): p. 1094.e1-1094.e5.
- 17. Colaneri, M., et al., *Lack of SARS-CoV-2 RNA environmental contamination in a tertiary referral hospital for infectious diseases in Northern Italy.* Journal of Hospital Infection, 2020. **105**(3): p. 474-476.
- 18. Mondelli, M.U., et al., *Low risk of SARS-CoV-2 transmission by fomites in real-life conditions*. The Lancet Infectious Diseases, 2020.
- 19. Hinds, W.C., *Aerosol technology: properties, behavior, and measurement of airborne particles.* 1999: John Wiley & Sons.
- 20. International Organization for Standardization (ISO), *ISO* 7708:1995(*E*) Air quality *Particle size fraction definitions for health-related sampling*. 1995.
- 21. Randall, K.a.E., E. Thomas and Marr, Linsey and Jimenez, Jose and Bourouiba, Lydia, , How Did We Get Here: What Are Droplets and Aerosols and How Far Do They Go? A Historical Perspective on the Transmission of Respiratory Infectious Diseases SSRN, 2021.
- 22. Stadnytskyi, V., P. Anfinrud, and A. Bax, *Breathing, speaking, coughing or sneezing: What drives transmission of SARS-CoV-2?* J Intern Med, 2021.
- 23. Liu, L., et al., *Short-range airborne transmission of expiratory droplets between two people*. Indoor Air, 2017. **27**(2): p. 452-462.
- 24. World Health Organization (WHO). *Transmission of SARS-CoV-2: implications for infection prevention precautions*. 2020; Available from: <u>https://www.who.int/news-room/commentaries/detail/transmission-of-sars-cov-2-implications-for-infection-prevention-precautions</u>.
- 25. Lu, J. and Z. Yang, *COVID-19 Outbreak Associated with Air Conditioning in Restaurant, Guangzhou, China, 2020.* Emerg Infect Dis, 2020. **26**(11): p. 2791-2793.
- 26. Shen, Y., et al., *Community Outbreak Investigation of SARS-CoV-2 Transmission Among Bus Riders in Eastern China.* JAMA Intern Med, 2020. **180**(12): p. 1665-1671.
- 27. Correia, G., et al., *Airborne route and bad use of ventilation systems as non-negligible factors in SARS-CoV-2 transmission.* Med Hypotheses, 2020. **141**: p. 109781.
- 28. Chu, D.K.W., et al., *SARS-CoV-2 Superspread in Fitness Center, Hong Kong, China, March 2021.* Emerging Infectious Disease journal, 2021. **27**(8).
- 29. Blocken, B., et al., *Ventilation and air cleaning to limit aerosol particle concentrations in a gym during the COVID-19 pandemic.* Build Environ, 2021. **193**: p. 107659.

- 30. Azimi, P., et al., *Mechanistic transmission modeling of COVID-19 on the Diamond Princess cruise ship demonstrates the importance of aerosol transmission.* Proc Natl Acad Sci U S A, 2021. **118**(8).
- 31. Li, Y., et al., *Probable airborne transmission of SARS-CoV-2 in a poorly ventilated restaurant*. Build Environ, 2021. **196**: p. 107788.
- 32. Chia, P.Y., et al., *Detection of air and surface contamination by SARS-CoV-2 in hospital rooms of infected patients*. Nat Commun, 2020. **11**(1): p. 2800.
- 33. Liu, Y., et al., *Aerodynamic analysis of SARS-CoV-2 in two Wuhan hospitals*. Nature, 2020. **582**(7813): p. 557-560.
- 34. Zhang, X.S. and C. Duchaine, *SARS-CoV-2 and Health Care Worker Protection in Low-Risk Settings: a Review of Modes of Transmission and a Novel Airborne Model Involving Inhalable Particles.* Clinical microbiology reviews, 2020. **34**(1): p. e00184-20.
- 35. Tang, S., et al., *Aerosol transmission of SARS-CoV-2? Evidence, prevention and control.* Environ Int, 2020. **144**: p. 106039.
- 36. Richard, M., et al., *SARS-CoV-2 is transmitted via contact and via the air between ferrets.* Nat Commun, 2020. **11**(1): p. 3496.
- 37. Kutter, J.S., et al., *SARS-CoV and SARS-CoV-2 are transmitted through the air between ferrets over more than one meter distance.* Nat Commun, 2021. **12**(1): p. 1653.
- 38. Leung, N.H.L., *Transmissibility and transmission of respiratory viruses*. Nat Rev Microbiol, 2021: p. 1-18.
- 39. De Wilde, A.H., et al., *Host Factors in Coronavirus Replication*. Curr Top Microbiol Immunol, 2018. **419**: p. 1-42.
- 40. Yi, Y., et al., *COVID-19: what has been learned and to be learned about the novel coronavirus disease.* Int J Biol Sci, 2020. **16**(10): p. 1753-1766.
- 41. Peeri, N.C., et al., *The SARS, MERS and novel coronavirus (COVID-19) epidemics, the newest and biggest global health threats: what lessons have we learned?* Int J Epidemiol, 2020. **49**(3): p. 717-726.
- 42. Bar-On, Y.M., et al., *SARS-CoV-2 (COVID-19) by the numbers*. eLife, 2020. **9**: p. e57309.
- 43. Elias, C., et al., *The incubation period of COVID-19: A meta-analysis*. Int J Infect Dis, 2021. **104**: p. 708-710.
- 44. Yan, R., et al., *Structural basis for the recognition of SARS-CoV-2 by full-length human ACE2*. Science, 2020. **367**(6485): p. 1444-1448.

- 45. Ortiz, M.E., et al., *Heterogeneous expression of the SARS-Coronavirus-2 receptor ACE2 in the human respiratory tract.* EBioMedicine, 2020. **60**: p. 102976.
- Wan, Y., et al., Receptor Recognition by the Novel Coronavirus from Wuhan: an Analysis Based on Decade-Long Structural Studies of SARS Coronavirus. J Virol, 2020. 94(7).
- 47. Nguyen, H.L., et al., *Does SARS-CoV-2 Bind to Human ACE2 More Strongly Than Does SARS-CoV?* J Phys Chem B, 2020. **124**(34): p. 7336-7347.
- 48. Peacock, T.P., et al., *SARS-CoV-2 one year on: evidence for ongoing viral adaptation*. J Gen Virol, 2021. **102**(4).
- 49. World Health Organization (WHO). *Naming and tracking SARS-CoV-2 variants of concern and variants of interest*. 2021; Available from: <u>https://www.who.int/news-room/events/detail/2021/06/03/default-calendar/naming-and-tracking-sars-cov-2-variants-of-concern-and-variants-of-interest</u>.
- 50. Washington, N.L., et al., *Genomic epidemiology identifies emergence and rapid* transmission of SARS-CoV-2 B.1.1.7 in the United States. medRxiv, 2021.
- 51. Volz, E., et al., *Assessing transmissibility of SARS-CoV-2 lineage B.1.1.7 in England.* Nature, 2021.
- 52. Takayama, K., *In Vitro and Animal Models for SARS-CoV-2 research*. Trends Pharmacol Sci, 2020. **41**(8): p. 513-517.
- 53. World Health Organization (WHO). *Cleaning and disinfection of environmental surfaces in the context of COVID-19. Interim guidance.* WHO/2019nCoV/Disinfection/2020.1 2020; Available from: <u>https://www.who.int/publications/i/item/cleaning-and-disinfection-of-environmental-</u> *surfaces-inthe-context-of-covid-19.*
- 54. Van Doremalen, N., et al., *Aerosol and Surface Stability of SARS-CoV-2 as Compared with SARS-CoV-1*. New England Journal of Medicine, 2020. **382**(16): p. 1564-1567.
- 55. Chan, K.H., et al., *The Effects of Temperature and Relative Humidity on the Viability of the SARS Coronavirus*. Adv Virol, 2011. **2011**: p. 734690.
- 56. Dabisch, P.S., M.; Herzog, A.; Beck, K.; Wood, S.; Krause, M.; Miller, D.; Weaver, W.; Freeburger, D.; Hooper, I.; Green, B.; Williams, G.; Holland, B.; Bohannon, J.; Wahl, V.; Yolitz, J.; Hevey, M.; Ratnesar-Shumate, S., *The influence of temperature, humidity, and simulated sunlight on the infectivity of SARS-CoV-2 in aerosols.* Aerosol Science and Technology, 2021. 55(2): p. 142-143.
- 57. Kampf, G., et al., *Potential sources, modes of transmission and effectiveness of prevention measures against SARS-CoV-2.* J Hosp Infect, 2020. **106**(4): p. 678-697.
- 58. Dowell, S.F., et al., *Severe acute respiratory syndrome coronavirus on hospital surfaces*. Clin Infect Dis, 2004. **39**(5): p. 652-7.

- Smither, S.J., et al., *Experimental aerosol survival of SARS-CoV-2 in artificial saliva and tissue culture media at medium and high humidity*. Emerg Microbes Infect, 2020. 9(1): p. 1415-1417.
- 60. Schuit, M., et al., *Airborne SARS-CoV-2 Is Rapidly Inactivated by Simulated Sunlight*. J Infect Dis, 2020. **222**(4): p. 564-571.
- 61. Ijaz, M.K., et al., *Survival characteristics of airborne human coronavirus 229E*. J Gen Virol, 1985. **66** (**Pt 12**): p. 2743-8.
- 62. Schuit, M., et al., *The stability of an isolate of the SARS-CoV-2 B.1.1.7 lineage in aerosols is similar to three earlier isolates.* J Infect Dis, 2021.
- 63. Majra, D., et al., *SARS-CoV-2 (COVID-19) superspreader events*. J Infect, 2021. **82**(1): p. 36-40.
- 64. Adam, D.C., et al., *Clustering and superspreading potential of SARS-CoV-2 infections in Hong Kong.* Nat Med, 2020. **26**(11): p. 1714-1719.
- 65. Bae, S., et al., *Transmission Characteristics of SARS-CoV-2 That Hinder Effective Control.* Immune Netw, 2021. **21**(1): p. e9.
- 66. Karimzadeh, S., R. Bhopal, and H. Nguyen Tien, *Review of infective dose, routes of transmission and outcome of COVID-19 caused by the SARS-COV-2: comparison with other respiratory viruses.* Epidemiol Infect, 2021. **149**: p. e96.
- 67. Cevik, M., et al., *SARS-CoV-2, SARS-CoV, and MERS-CoV viral load dynamics, duration of viral shedding, and infectiousness: a systematic review and meta-analysis.* Lancet Microbe, 2021. **2**(1): p. e13-e22.
- 68. Atkinson, B. and E. Petersen, *SARS-CoV-2 shedding and infectivity*. Lancet, 2020. **395**(10233): p. 1339-1340.
- 69. Kim, M.C., et al., *Duration of Culturable SARS-CoV-2 in Hospitalized Patients with Covid-19.* N Engl J Med, 2021. **384**(7): p. 671-673.
- 70. Vaught, E.S., et al., *Prolonged Sars-CoV-2 Viral Culture Detected in a Mild COVID-19 Infection*. Mil Med, 2021.
- 71. Qiu, X., et al., *The role of asymptomatic and pre-symptomatic infection in SARS-CoV-2 transmission-a living systematic review*. Clin Microbiol Infect, 2021. **27**(4): p. 511-519.
- 72. Oran, D.P. and E.J. Topol, *Prevalence of Asymptomatic SARS-CoV-2 Infection : A Narrative Review*. Ann Intern Med, 2020. **173**(5): p. 362-367.
- Pan, Y., et al., *Viral load of SARS-CoV-2 in clinical samples*. Lancet Infect Dis, 2020.
 20(4): p. 411-412.

- Peiris, J.S., et al., *Clinical progression and viral load in a community outbreak of coronavirus-associated SARS pneumonia: a prospective study.* Lancet, 2003. 361(9371): p. 1767-72.
- 75. Avadhanula, V., et al., Viral load of SARS-CoV-2 in adults during the first and second wave of COVID-19 pandemic in Houston, TX: the potential of the super-spreader. J Infect Dis, 2021.
- 76. Wolfel, R., et al., *Virological assessment of hospitalized patients with COVID-2019*. Nature, 2020. **581**(7809): p. 465-469.
- 77. Milton, D.K., et al., *Influenza virus aerosols in human exhaled breath: particle size, culturability, and effect of surgical masks.* PLoS Pathog, 2013. **9**(3): p. e1003205.
- 78. Lednicky, J.A., et al., *Viable SARS-CoV-2 in the air of a hospital room with COVID-19 patients*. Int J Infect Dis, 2020. **100**: p. 476-482.
- 79. Liu, L., et al., *Evaporation and dispersion of respiratory droplets from coughing*. Indoor Air, 2017. **27**(1): p. 179-190.
- 80. Åkervik, E., et al., *Luftbåren smitte av virale luftveisinfeksjoner fra et aerorsolfysisk perspektiv.* 2020, Forsvarets Forskningsinstitutt.
- 81. Duguid, J.P., *The size and the duration of air-carriage of respiratory droplets and droplet-nuclei.* J Hyg (Lond), 1946. **44**(6): p. 471-9.
- 82. Nicas, M., W.W. Nazaroff, and A. Hubbard, *Toward understanding the risk of secondary airborne infection: emission of respirable pathogens*. J Occup Environ Hyg, 2005. **2**(3): p. 143-54.
- 83. Johnson, G.R., et al., *Modality of human expired aerosol size distributions*. Journal of Aerosol Science, 2011. **42**(12): p. 839-851.
- 84. Lieber, C., et al., *Insights into the evaporation characteristics of saliva droplets and aerosols: Levitation experiments and numerical modeling.* J Aerosol Sci, 2021. **154**: p. 105760.
- 85. Wells, H.F., *On Air-borne Infection. Study II. Droplets and Droplet Nuclei.* American Journal of Hygiene, 1934. **20**: p. 611-618.
- 86. Mittal, R., R. Ni, and J.-H. Seo, *The flow physics of COVID-19*. Journal of fluid Mechanics, 2020. **894**.
- 87. Chao, C.Y.H., et al., *Characterization of expiration air jets and droplet size distributions immediately at the mouth opening*. J Aerosol Sci, 2009. **40**(2): p. 122-133.
- 88. Asadi, S., et al., *Aerosol emission and superemission during human speech increase with voice loudness.* Sci Rep, 2019. **9**(1): p. 2348.

- Chao, F.L., *Face mask designs following novel Coronavirus*. J Public Health Res, 2020. 9(1): p. 1770.
- 90. Loudon, R.G. and R.M. Roberts, *Droplet Expulsion from the Respiratory Tract.* American Review of Respiratory Disease, 1966. **95**(3): p. 435–442.
- 91. Papineni, R.S. and F.S. Rosenthal, *The size distribution of droplets in the exhaled breath of healthy human subjects.* J Aerosol Med, 1997. **10**(2): p. 105-16.
- 92. Fabian, P., et al., *Influenza virus in human exhaled breath: an observational study*. PLoS One, 2008. **3**(7): p. e2691.
- 93. Lindsley, W.G., et al., *Quantity and Size Distribution of Cough-Generated Aerosol Particles Produced by Influenza Patients During and After Illness.* Journal of Occupational and Environmental Hygiene, 2012. **9**(7): p. 443-449.
- 94. Asadi, S., et al., *Efficacy of masks and face coverings in controlling outward aerosol particle emission from expiratory activities.* Sci Rep, 2020. **10**(1): p. 15665.
- 95. Fennelly, K.P., *Particle sizes of infectious aerosols: implications for infection control.* Lancet Respir Med, 2020. **8**(9): p. 914-924.
- 96. Lee, B.U., *Minimum Sizes of Respiratory Particles Carrying SARS-CoV-2 and the Possibility of Aerosol Generation.* Int J Environ Res Public Health, 2020. **17**(19).
- 97. Park, W.B., et al., *Virus Isolation from the First Patient with SARS-CoV-2 in Korea.* J Korean Med Sci, 2020. **35**(7): p. e84.
- 98. Kim, J.M., et al., *Identification of Coronavirus Isolated from a Patient in Korea with COVID-19*. Osong Public Health Res Perspect, 2020. **11**(1): p. 3-7.
- 99. Bourouiba, L., *The fluid dynamics of disease transmission*. Annual Review of Fluid Mechanics, 2020. **53**.
- 100. Bourouiba, L., E. Dehandschoewercker, and J.W.M. Bush, *Violent expiratory events:* on coughing and sneezing. Journal of Fluid Mechanics, 2014. **745**: p. 537-563.
- 101. Abkarian, M., et al., *Speech can produce jet-like transport relevant to asymptomatic spreading of virus.* Proceedings of the National Academy of Sciences, 2020. **117**(41): p. 25237-25245.
- 102. Xie, X., et al., *How far droplets can move in indoor environments–revisiting the Wells evaporation–falling curve.* Indoor air, 2007. **17**(3): p. 211-225.
- Bourouiba, L., Turbulent gas clouds and respiratory pathogen emissions: potential implications for reducing transmission of COVID-19. Jama, 2020. 323(18): p. 1837-1838.
- 104. Prasanna Simha, P. and P.S. Mohan Rao, *Universal trends in human cough airflows at large distances*. Phys Fluids (1994), 2020. **32**(8): p. 081905.

- 105. Viola, I.M., et al., *Face Coverings, Aerosol Dispersion and Mitigation of Virus Transmission Risk.* IEEE Open Journal of Engineering in Medicine and Biology, 2021.
 2: p. 26-35.
- 106. Abkarian, M., et al., *Speech can produce jet-like transport relevant to asymptomatic spreading of virus.* Proceedings of the National Academy of Sciences, 2020. **117**(41): p. 25237.
- 107. Yang, F., et al., *Towards improved social distancing guidelines: Space and time dependence of virus transmission from speech-driven aerosol transport between two individuals.* Physical Review Fluids, 2020. **5**(12): p. 122501.
- Bazant, M.Z. and J.W.M. Bush, A guideline to limit indoor airborne transmission of COVID-19. Proceedings of the National Academy of Sciences, 2021. 118(17): p. e2018995118.
- 109. Johansson, M.A., et al., *SARS-CoV-2 Transmission From People Without COVID-19 Symptoms*. JAMA Network Open, 2021. **4**(1): p. e2035057-e2035057.
- 110. Cichowicz, J.K., M. Casey, and M.M. D'Alessandro. *Respiratory Protection vs. Source Control What's the difference?* CDC blog 2020; Available from: https://blogs.cdc.gov/niosh-science-blog/2020/09/08/source-control/.
- Lee, K.W. and B.Y.H. Liu, On the Minimum Efficiency and the Most Penetrating Particle Size for Fibrous Filters. Journal of the Air Pollution Control Association, 1980. 30(4): p. 377-381.
- 112. Da Roza, R.A., *Particle size for greatest penetration of HEPA filters and their true efficiency*. 1982, Lawrence Livermore National Lab., CA (USA). p. Medium: ED; Size: Pages: 17.
- 113. Xiao, H.M., G.J. Chen, and Y.P. Song, *Penetration Performance of Melt-Blown Polypropylene Electret Nonwoven Web against DEHS Aerosols*. Advanced Materials Research, 2012. **393-395**: p. 1318-1321.
- 114. He, X., et al., *Effect of Particle Size on the Performance of an N95 Filtering Facepiece Respirator and a Surgical Mask at Various Breathing Conditions*. Aerosol Science and Technology, 2013. **47**.
- 115. European Committee For Standardization (CEN), *EN 149:2001+A1 Respiratory* protective devices - Filtering half masks to protect against particles - Requirements, testing, marking. 2001.
- European Committee For Standardization (CEN), Respiratory protective devices -Methods of test - Part 1: Determination of inward leakage and total inward leakage. 2001.
- 117. Occupational Safety and Health Administration (OSHA), Occupational Safety and Health Standards OSHA 3352-02 Assigned Protection Factors for the Revised Respiratory Protection Standard. 2009.

- 118. European Committee For Standardization (CEN), *EN 529:2005 Respiratory protective devices Recommendations for selection, use, care and maintenance Guidance document.* 2005.
- 119. Wen, Z., et al., *Determining the filtration efficiency of half-face medical protection* mask (N99) against viral aerosol. Aerobiologia (Bologna), 2010. **26**(3): p. 245-251.
- 120. Rengasamy, S., et al., *A comparison of facemask and respirator filtration test methods*. Journal of Occupational and Environmental Hygiene, 2017. **14**(2): p. 92-103.
- 121. National Institute for Occupational Safety and Health (NIOSH), Determination of particulate filter efficiency level for N95 series filters against solid particulates for non-powered air-purifying respirators standard testing procedure 2019.
- 122. American Society for Testing and Materials (ASTM), F2101-19 Standard Test Method for Evaluating the Bacterial Filtration Efficiency (BFE) of Medical Face Mask Materials, Using a Biological Aerosol of Staphylococcus aureus. 2019.
- 123. Bałazy, A., et al., *Do N95 respirators provide 95% protection level against airborne viruses, and how adequate are surgical masks?* American Journal of Infection Control, 2006. **34**(2): p. 51-57.
- 124. Rengasamy, A., Z. Zhuang, and R. BerryAnn, *Respiratory protection against bioaerosols: Literature review and research needs*. American Journal of Infection Control, 2004. **32**(6): p. 345-354.
- 125. Tamayol, A., K.W. Wong, and M. Bahrami, *Effects of microstructure on flow properties of fibrous porous media at moderate Reynolds number*. Phys Rev E Stat Nonlin Soft Matter Phys, 2012. **85**(2 Pt 2): p. 026318.
- 126. Xi, J., X.A. Si, and R. Nagarajan, *Effects of mask-wearing on the inhalability and deposition of airborne SARS-CoV-2 aerosols in human upper airway.* Physics of Fluids, 2020. **32**(12): p. 123312.
- 127. Perić, R. and M. Perić, *Analytical and Numerical Investigation of the Airflow in Face Masks used for Protection against COVID-19 Virus –Implications for Mask Design and Usage*. Journal of Applied Fluid Mechanics, 2020. **13**(06).
- 128. International Organization for Standardization (ISO), *ISO/TS 16975-1:2016 Respiratory* protective devices — Selection, use and maintenance — Part 1: Establishing and implementing a respiratory protective device programme, *ISO/TC 94/SC 15 Respiratory* protective devices. 2016.
- 129. Tønsager, J., Desinfeksjon og gjenbruk av åndedrettsvern (filtrerende halvmasker) litteratursammenstilling, FFI, Editor. 2020, Forsvarets Forskningsinstitutt.
- Rengasamy, S., B.C. Eimer, and R.E. Shaffer, Evaluation of the Performance of the N95-Companion: Effects of Filter Penetration and Comparison with Other Aerosol Instruments. Journal of Occupational and Environmental Hygiene, 2012. 9(7): p. 417-426.
- 131. 3M. *Technical Data Bulletin #171: Nanotechnology and Respirator Use*. 2015; Available from: <u>https://multimedia.3m.com/mws/media/376179O/nanotechnology-and-respirator-use.pdf</u>.
- 132. Rengasamy, S., B.C. Eimer, and J. Szalajda, A Quantitative Assessment of the Total Inward Leakage of NaCl Aerosol Representing Submicron-Size Bioaerosol Through N95 Filtering Facepiece Respirators and Surgical Masks. Journal of Occupational and Environmental Hygiene, 2014. **11**(6): p. 388-396.
- 133. Ueki, H., et al., *Effectiveness of Face Masks in Preventing Airborne Transmission of SARS-CoV-2*. mSphere, 2020. **5**(5).
- 134. Lee, S.A., S.A. Grinshpun, and T. Reponen, *Respiratory performance offered by N95* respirators and surgical masks: human subject evaluation with NaCl aerosol representing bacterial and viral particle size range. Ann Occup Hyg, 2008. **52**(3): p. 177-85.
- 135. Reponen, T., et al., *Effect of fit testing on the protection offered by n95 filtering facepiece respirators against fine particles in a laboratory setting.* Ann Occup Hyg, 2011. **55**(3): p. 264-71.
- 136. Rengasamy, S., et al., A comparison of total inward leakage measured using sodium chloride (NaCl) and corn oil aerosol methods for air-purifying respirators. J Occup Environ Hyg, 2018. **15**(8): p. 616-627.
- 137. Rengasamy, S., et al., *Total inward leakage measurement of particulates for N95 filtering facepiece respirators--a comparison study*. Ann Occup Hyg, 2014. **58**(2): p. 206-16.
- 138. He, X., et al., How Does Breathing Frequency Affect the Performance of an N95 Filtering Facepiece Respirator and a Surgical Mask Against Surrogates of Viral Particles? Journal of Occupational and Environmental Hygiene, 2014. 11(3): p. 178-185.
- 139. He, X., et al., *Effects of breathing frequency and flow rate on the total inward leakage of an elastomeric half-mask donned on an advanced manikin headform.* Ann Occup Hyg, 2014. **58**(2): p. 182-94.
- 140. European Union (EU). *Technical documentation and EU declaration of conformity*. 2021; Available from: <u>https://europa.eu/youreurope/business/product-requirements/compliance/technical-documentation-conformity/index_en.htm</u>.
- 141. Statens Legemiddelverk (SLV). *Placing on the market of medical devices Procedure and responsibility related to CE marking and placing on the market of medical devices.* 2021; Available from: <u>https://legemiddelverket.no/english/medical-devices/regulatory-information-regarding-medical-devices/placing-on-the-market-of-medical-devices.</u>
- 142. American Society for Testing and Materials (ASTM), F2299/F2299M-03 Standard Test Method for Determining the Initial Efficiency of Materials Used in Medical Face Masks to Penetration by Particulates Using Latex Spheres. 2017.

- 143. American Society for Testing and Materials (ASTM), F1862/F1862M-17 Standard Test Method for Resistance of Medical Face Masks to Penetration by Synthetic Blood (Horizontal Projection of Fixed Volume at a Known Velocity). 2017.
- 144. American Society for Testing and Materials (ASTM), F2100-19 Standard Specification for Performance of Materials Used in Medical Face Masks. 2019.
- 145. European Committee For Standardization (CEN), *EN 14683:2019+AC:2019 Medical face masks Requirements and test methods.* 2019.
- 146. Chen, C. and K. Willeke, *Aerosol penetration through surgical masks*. American journal of infection control, 1992. **20 4**: p. 177-84.
- 147. Oberg, T. and L. Brosseau, *Surgical mask filter and fit performance*. American Journal of Infection Control, 2008. **36**: p. 276 282.
- 148. Clapp, P.W., et al., Evaluation of Cloth Masks and Modified Procedure Masks as Personal Protective Equipment for the Public During the COVID-19 Pandemic. JAMA Intern Med, 2020.
- 149. Whiley, H., et al., Viral Filtration Efficiency of Fabric Masks Compared with Surgical and N95 Masks. Pathogens, 2020. **9**(9).
- 150. European Committee For Standardization (CEN), *SN-CWA 17553:2020 Workshop Agreement, Community face coverings - Guide to minimum requirements, methods of testing and use.* 2020.
- 151. French Standardization Association (AFNOR), SPEC S76-001 Barrier mask Guide to minimum requirements, methods of testing, making and use. 2020.
- 152. Royal Netherlands Standardization Institute (NEN), NCS 1-2 (en) Certification Scheme Community Masks - Conformity assessment - Requirements for certifying community masks according to NEN-spec 1-2:2020-11-09. 2020, NEN.
- 153. Barret, A. *COVID-19: Face shields alone 'unlikely' to protect hairdressers and barbers*. 2020; Available from: <u>https://www.sciencefocus.com/news/covid-19-face-shields-alone-unlikely-to-protect-hairdressers-and-barbers/</u>.
- 154. The Connexion. *Covid-19 France: Fines for wearing plastic face visors*. 2020; Available from: <u>https://www.connexionfrance.com/French-news/Covid-19-France-Fines-for-wearing-plastic-face-visors-instead-of-masks</u>.
- 155. Napoli, P.E., et al., *The Ocular Surface and the Coronavirus Disease 2019: Does a Dual 'Ocular Route' Exist?* Journal of Clinical Medicine, 2020. **9**(5): p. 1269.
- 156. Bischoff, W.E., et al., *Transocular Entry of Seasonal Influenza–Attenuated Virus Aerosols and the Efficacy of N95 Respirators, Surgical Masks, and Eye Protection in Humans.* The Journal of Infectious Diseases, 2011. **204**(2): p. 193-199.

- 157. Coroneo, M.T. and P.J. Collignon, *SARS-CoV-2: eye protection might be the missing key*. The Lancet Microbe, 2021.
- 158. Coroneo, M.T., *The eye as the discrete but defensible portal of coronavirus infection*. The Ocular Surface, 2021. **19**: p. 176-182.
- 159. Perencevich, E.N., D.J. Diekema, and M.B. Edmond, *Moving Personal Protective Equipment Into the Community: Face Shields and Containment of COVID-19.* JAMA, 2020. **323**(22): p. 2252-2253.
- 160. Verma, S., M. Dhanak, and J. Frankenfield, *Visualizing droplet dispersal for face shields and masks with exhalation valves.* Physics of Fluids, 2020. **32**(9): p. 091701.
- 161. Lindsley, W.G., et al., *Efficacy of face shields against cough aerosol droplets from a cough simulator.* J Occup Environ Hyg, 2014. **11**(8): p. 509-18.
- 162. Roberge, R.J., *Face shields for infection control: A review.* Journal of Occupational and Environmental Hygiene, 2016. **13**(4): p. 235-242.
- 163. European Committee For Standardization (CEN), *EN 166:2001 Personal eye-protection* - *Specifications*. 2001.
- 164. European Committee For Standardization (CEN), *EN 168:2001 Personal eye-protection* - *Non-optical test methods.* 2001.
- 165. European Committee For Standardization (CEN), *EN 167:2001 Personal eye-protection* - *Optical test methods.* 2001.
- 166. Schilling, K., et al., *An accessible method for screening aerosol filtration identifies poor-performing commercial masks and respirators*. J Expo Sci Environ Epidemiol, 2020: p. 1-10.
- 167. Mueller, A.V., et al., *Quantitative Method for Comparative Assessment of Particle Filtration Efficiency of Fabric Masks as Alternatives to Standard Surgical Masks for PPE*. medRxiv, 2020: p. 2020.04.17.20069567.
- Xiao, L.I., H. Sakagami, and N. Miwa, A New Method for Testing Filtration Efficiency of Mask Materials Under Sneeze-like Pressure. In Vivo, 2020. 34(3 Suppl): p. 1637-1644.
- 169. Drewnick, F., et al., Aerosol filtration efficiency of household materials for homemade face masks: Influence of material properties, particle size, particle electrical charge, face velocity, and leaks. Aerosol Science and Technology, 2021. **55**(1): p. 63-79.
- 170. Tuomi, T., *Face seal leakage of half masks and surgical masks*. Am Ind Hyg Assoc J, 1985. **46**(6): p. 308-12.
- 171. Weber, A., et al., *Aerosol penetration and leakae characteristics of masks used in the health care industry*. American Journal of Infection Control, 1993. **21**(4): p. 167-173.

- 172. Clase, C.M., et al., *Forgotten Technology in the COVID-19 Pandemic: Filtration Properties of Cloth and Cloth Masks—A Narrative Review.* Mayo Clinic Proceedings, 2020. **95**(10): p. 2204-2224.
- 173. Jain, M., et al., *Efficacy and Use of Cloth Masks: A Scoping Review*. Cureus, 2020. 12(9): p. e10423.
- 174. Konda, A., et al., *Aerosol Filtration Efficiency of Common Fabrics Used in Respiratory Cloth Masks.* ACS Nano, 2020. **14**(5): p. 6339-6347.
- 175. Rengasamy, S., B. Eimer, and R.E. Shaffer, *Simple respiratory protection--evaluation* of the filtration performance of cloth masks and common fabric materials against 20-1000 nm size particles. Ann Occup Hyg, 2010. **54**(7): p. 789-98.
- 176. Zangmeister, C.D., et al., Filtration Efficiencies of Nanoscale Aerosol by Cloth Mask Materials Used to Slow the Spread of SARS-CoV-2. ACS Nano, 2020. 14(7): p. 9188-9200.
- 177. Wang, Y., et al., *Reduction of secondary transmission of SARS-CoV-2 in households by face mask use, disinfection and social distancing: a cohort study in Beijing, China.* BMJ Global Health, 2020. **5**(5): p. e002794.
- 178. Furuhashi, M., A study on the microbial filtration efficiency of surgical face masks-with special reference to the non-woven fabric mask. Bull Tokyo Med Dent Univ, 1978.
 25(1): p. 7-15.
- 179. Wang, D., et al., Selection of homemade mask materials for preventing transmission of COVID-19: A laboratory study. PLoS One, 2020. **15**(10): p. e0240285.
- 180. Verma, S., M. Dhanak, and J. Frankenfield, *Visualizing the effectiveness of face masks in obstructing respiratory jets.* Physics of Fluids, 2020. **32**(6): p. 061708.
- 181. Kähler, C.J. and R. Hain, *Fundamental protective mechanisms of face masks against droplet infections*. Journal of aerosol science, 2020. **148**: p. 105617.
- 182. Tang, J.W., et al., *A schlieren optical study of the human cough with and without wearing masks for aerosol infection control.* Journal of the Royal Society Interface, 2009. **6**(suppl_6): p. S727-S736.
- 183. Dbouk, T. and D. Drikakis, *On respiratory droplets and face masks*. Physics of Fluids, 2020. **32**(6): p. 063303.
- 184. Pendar, M.-R. and P.J. C, *Numerical modeling of the distribution of virus carrying saliva droplets during sneeze and cough.* Phys Fluids, 2020. **32**(6): p. 061708.
- 185. Khosronejad, A., et al., *Fluid dynamics simulations show that facial masks can suppress the spread of COVID-19 in indoor environments.* Aip Advances, 2020. **10**(12).

- 186. Lei, Z., J.J. Yang, and Z. Zhuang, *Headform and N95 filtering facepiece respirator interaction: contact pressure simulation and validation*. J Occup Environ Hyg, 2012. 9(1): p. 46-58.
- 187. Dowell, E.H. and K.C. Hall, *Modeling of fluid-structure interaction*. Annual Review of Fluid Mechanics, 2001. **33**(1): p. 445-490.
- Lei, Z., et al., Simulation and evaluation of respirator faceseal leaks using computational fluid dynamics and infrared imaging. Ann Occup Hyg, 2013. 57(4): p. 493-506.
- 189. Xie, X., et al., *Exhaled droplets due to talking and coughing*. J R Soc Interface, 2009. 6 Suppl 6: p. S703-14.
- 190. Bandiera, L., et al., *Face coverings and respiratory tract droplet dispersion*. R Soc Open Sci, 2020. **7**(12): p. 201663.
- 191. Qian, Y., et al., *Particle Reentrainment from Fibrous Filters*. Aerosol Science and Technology, 1997. **27**(3): p. 394-404.
- 192. Kolewe, E.L., et al., *Check the gap: Facemask performance and exhaled aerosol distributions around the wearer*. PLOS ONE, 2020. **15**(12): p. e0243885.
- 193. Diaz, K.T. and G.C. Smaldone, *Quantifying exposure risk: surgical masks and respirators*. Am J Infect Control, 2010. **38**(7): p. 501-8.
- 194. Mansour, M.M. and G.C. Smaldone, *Respiratory source control versus receiver* protection: impact of facemask fit. J Aerosol Med Pulm Drug Deliv, 2013. **26**(3): p. 131-7.
- 195. Patel, R.B., et al., *Respiratory source control using a surgical mask: An in vitro study*. Journal of Occupational and Environmental Hygiene, 2016. **13**(7): p. 569-576.
- 196. Brooks, J.T., et al., *Maximizing Fit for Cloth and Medical Procedure Masks to Improve Performance and Reduce SARS-CoV-2 Transmission and Exposure, 2021.* MMWR. Morbidity and Mortality Weekly Report, 2021. **70**(7): p. 1-4.
- 197. Matuschek, C., et al., *Face masks: benefits and risks during the COVID-19 crisis*. Eur J Med Res, 2020. **25**(1): p. 32.
- 198. Jefferson, T., et al., *Physical interventions to interrupt or reduce the spread of respiratory viruses.* Cochrane Database of Systematic Reviews, 2020(11).
- 199. Wei, J., et al., Why does the spread of COVID-19 vary greatly in different countries? Revealing the efficacy of face masks in epidemic prevention. Epidemiol Infect, 2021.
 149: p. e24.
- 200. Shakya, K.M., et al., *Evaluating the efficacy of cloth facemasks in reducing particulate matter exposure.* J Expo Sci Environ Epidemiol, 2017. **27**(3): p. 352-357.

- 201. Lima, M.M.S., et al., *Cloth face masks to prevent Covid-19 and other respiratory infections*. Rev Lat Am Enfermagem, 2020. **28**: p. e3353.
- 202. Davies, A., et al., *Testing the Efficacy of Homemade Masks: Would They Protect in an Influenza Pandemic?* Disaster Medicine and Public Health Preparedness, 2013. **7**(4): p. 413-418.
- 203. Loeb, M., et al., Surgical Mask vs N95 Respirator for Preventing Influenza Among Health Care Workers: A Randomized Trial. JAMA, 2009. **302**(17): p. 1865-1871.
- 204. Lee, S.A., et al., *Particle Size-Selective Assessment of Protection of European Standard FFP Respirators and Surgical Masks against Particles-Tested with Human Subjects.* J Healthc Eng, 2016. **2016**.
- 205. Grinshpun, S.A., et al., *Performance of an N95 Filtering Facepiece Particulate Respirator and a Surgical Mask During Human Breathing: Two Pathways for Particle Penetration.* Journal of Occupational and Environmental Hygiene, 2009. **6**(10): p. 593-603.
- 206. Seo, H., J.I. Kim, and H. Kim, *Development of Korean Head forms for Respirator Performance Testing.* Saf Health Work, 2020. **11**(1): p. 71-79.
- 207. Huh, Y.J., et al., *Fit Characteristics of N95 Filtering Facepiece Respirators and the Accuracy of the User Seal Check among Koreans*. Infect Control Hosp Epidemiol, 2018. **39**(1): p. 104-107.
- 208. Oestenstad, R.K. and A.A. Bartolucci, *Factors Affecting the Location and Shape of Face Seal Leak Sites on Half-Mask Respirators*. Journal of Occupational and Environmental Hygiene, 2010. **7**(6): p. 332-341.
- 209. Oestenstad, R.K., H.K. Dillion, and L.L. Perkins, *Distribution of Faceseal Leak Sites on a Half-Mask Respirator and Their Association with Facial Dimensions*. American Industrial Hygiene Association Journal, 1990. **51**(5): p. 285-290.
- 210. Chen, C.C. and K. Willeke, *Characteristics of face seal leakage in filtering facepieces*. Am Ind Hyg Assoc J, 1992. **53**(9): p. 533-9.
- 211. Rengasamy, S. and B. Eimer, *Total inward leakage of nanoparticles through filtering facepiece respirators.* The Annals of occupational hygiene, 2011. **55 3**: p. 253-63.
- 212. Zhuang, Z., B. Bradtmiller, and R.E. Shaffer, *New respirator fit test panels representing the current U.S. civilian work force.* J Occup Environ Hyg, 2007. **4**(9): p. 647-59.
- 213. Sickbert-Bennett, E.E., et al., *Filtration Efficiency of Hospital Face Mask Alternatives Available for Use During the COVID-19 Pandemic*. JAMA Internal Medicine, 2020.
 180(12): p. 1607-1612.
- 214. Runde, D.P., et al., *The "double eights mask brace" improves the fit and protection of a basic surgical mask amidst COVID-19 pandemic.* J Am Coll Emerg Physicians Open, 2021. **2**(1): p. e12335.

- 215. Zangmeister, C.D., et al., *Hydration of Hydrophilic Cloth Face Masks Enhances the Filtration of Nanoparticles*. ACS Applied Nano Materials, 2021.
- 216. Kasloff, S.B., et al., *Stability of SARS-CoV-2 on critical personal protective equipment*. Sci Rep, 2021. **11**(1): p. 984.
- 217. Hao, W., et al., *Filtration performances of non-medical materials as candidates for manufacturing facemasks and respirators*. Int J Hyg Environ Health, 2020. **229**: p. 113582.
- 218. Neupane, B.B., et al., *Optical microscopic study of surface morphology and filtering efficiency of face masks.* PeerJ, 2019. **7**: p. e7142-e7142.

A Appendix

A.1 List of symbols and abbreviations

The list is given in alphabetical order.

ACE2	Angiotensin I converting enzyme 2	
AFNOR	French Standardization Association	
ASTM	American Society for Testing and Materials	
APF	Assigned Protection Factor	
BFE	Bacterial filtration efficiency	
CDC	U.S. Centers for Disease Control and Prevention	
CEN	European Committee for Standardization	
CFD	Computational fluid dynamics	
CFM	Community face mask	
CWA	CEN Workshop Agreement	
COVID-19	Coronavirus disease 2019	
ECDC	European Centre for Disease Prevention and Control	
EU	European Union	
FDA	US Food and Drug Administration	
FE	Filtration efficiency	
FF	Fit Factor	
FFE	Fitted Filtration Efficiency	
FFR	Filtering facepiece respirator	
HCW	Health care worker	
ISO	International Organization for Standardization	
MD	Medical Device	
MFM	Medical face mask	
NEN	Royal Netherlands Standardization Institute	
OSHA	Occupational Safety and Health Administration	
PAPR	Powered air-purifying respirator	
PF	Protection Factor	
PFE	Particle filtration efficiency	
PHEIC	Public health emergency of international concern	

PPE	Personal protective equipment
RPD	Respiratory protection device
SARS-CoV-2	Severe acute respiratory syndrome coronavirus 2
SCBA	Self-contained breathing apparatus
TIL	Total Inward Leakage
VFE	Viral filtration efficiency
WHO	World Health Organization

A.2 Terms and definitions

Fit factor, FF is calculated based on formula:

$$FF(\%) = \frac{C_{out}}{C_r} \times 100 \tag{A.1}$$

where C_{out} is the test aerosol concentration in the surroundings, C_r – test aerosol concentration inside the respirator

Total Inward Leakage TIL is calculated according to the formula:

$$TIL(\%) = \frac{P_1 + P_2 + \dots + P_n}{n} \times 100\%$$
(A.2)

where P_n is FFR penetration by the test aerosol during n-exercise:

$$P_n(\%) = \frac{C_{r,n}}{C_{out,n}} \times 100\%$$
 (A.3)

where $C_{r,n}$ – concentration of test aerosol inside the respirator during exercise n, $C_{out,n}$ – concentration of test aerosol during exercise n.

			requirement						
			Max. Filter penetration against		Aerosol partic median d	ol particle size (count nedian diameter)		Breathing resistance [mbar]	
standard	respirator o	class	solid aerosol (NaCl)	liquid aerosol (EN paraffin oil, US dioctyl phtalate DOP oil)	solid	liquid	Challenge flow	inhalation	exhalation
EN EN149	FFP1 FFP2 FFP3		80%	80%	0,06-0.1 μm	0.29-0.45 μm	95L/min	0.6 @ 30 L/min; 2.1 @ 95 L/min	3.0 @160 L/min
			94%	94%				0.7 @ 30 L/min; 2.4 @ 95 L/min	
			99%	99%				1.0 @ 30 L/min; 3.0 @ 95 L/min	
US NIOSH- 42 CFR 84	N-class	N95/99/100	95/99/99.97%	n/a	0.075±0.020 μm	n/a	95 I /min	3.43 @85	2.45 @85
	R-class	R95/99/100	n/a	95/99/99.97%	n/a	0.185±0.020	05 L/ MIN	L/min	L/min
	P-class	P95/99/100	n/a	95/99/99.97%	n/a	0.185±0.020			

 Table A.1
 Chosen requirements for FFR filtration efficiency tests according to European and US/international standards.

Table A.2A comparison of TIL procedures given in EU and US standards used for FFR test and evaluation.

Table A.2A comparison of TIL procedures given in EU and US standards used for FFR test
and evaluation.

EN 149 (Europe)	NIOSH/OSHA (US)
NaCl aerosol	NaCl or oil aerosol
walking for 2 min without head movement or	While standing: eight exercises performed in
talking;	the following order:
2) turning head from side to side	1) normal breathing, 2) deep breathing, 3)
(approximately 15 times), as if inspecting the	turn head side to side, 4) move head up and
walls of a tunnel for 2 min;	down, 5) speak out loud (recitation of the
3) moving the head up and down	'rainbow' passage), 6) reach for floor and
(approximately 15 times), as if inspecting the	ceiling, 7) grimace, and 8) normal breathing.
roof and floor for 2 min;	The duration of time for each exercise is
4) reciting the alphabet or an agreed text out	approximately one minute for a total of eight
loud as if communicating with a colleague	minutes for the test.
for 2 min;	
5) walking for 2 min without head movement	
or talking.	
at least 8 out of the 10 individual wearer	n/a*
arithmetic means for the total inward leakage	
shall be not greater than	
22% for FFP1	
8% for FFP2	
2% for FFP3.	

*NIOSH does not provide any specific numbers, but the respirator performance should comply with the legal national regulations in the US (42 Code of Federal Regulations §84).

 Table A.3
 List of chosen European and international standards relevant for MFM testing.

Standards	Comments
EN 14683:2019+AC:2019 Medical face masks -	European standard. Used to
Requirements and test methods	assess BFE, pressure drop
	across the mask material, for
	blood penetration refers to ISO
	22609
ISO 22609:2004 Clothing for protection against infectious	International. Used to assess
agents — Medical face masks — Test method for	blood penetration
resistance against penetration by synthetic blood (fixed	
volume, horizontally projected)	
ASTM F2100-19 Standard Specification for Performance	International (US).
of Materials Used in Medical Face Masks	Summarizes specifications and
	terminology used in F2299,
	F2101 and F1862

ASTM F2299/F2299M-03(2017) Standard Test Method	International (US). Used to		
for Determining the Initial Efficien	assess PFE of the medical face		
cy of Materials Used in Medical Face Masks to	mask material		
Penetration by Particulates Using Latex Spheres			
ASTM F2101-19 Standard Test Method for Evaluating	International (US). Used to		
the Bacterial Filtration Efficiency (BFE) of Medical Face	assess BFE		
Mask Materials, Using a Biological Aerosol of			
Staphylococcus aureus			
Mil-M369454C	US. Used to assess BFE		
ASTM F1862/F1862M-17 Standard Test Method for	International (US). Used to		
Resistance of Medical Face Masks to Penetration by	assess blood penetration.		
Synthetic Blood (Horizontal Projection of Fixed Volume	_		
at a Known Velocity)			
MIL-M-36954C Differential Pressure Air Permeability -	US. Used to assess pressure		
Respiratory Resistance Delta P Test	drop across mask material.		

About FFI

The Norwegian Defence Research Establishment (FFI) was founded 11th of April 1946. It is organised as an administrative agency subordinate to the Ministry of Defence.

FFI's mission

FFI is the prime institution responsible for defence related research in Norway. Its principal mission is to carry out research and development to meet the requirements of the Armed Forces. FFI has the role of chief adviser to the political and military leadership. In particular, the institute shall focus on aspects of the development in science and technology that can influence our security policy or defence planning.

FFI's vision

FFI turns knowledge and ideas into an efficient defence.

FFI's characteristics

Creative, daring, broad-minded and responsible.



Forsvarets forskningsinstitutt Postboks 25 2027 Kjeller

Besøksadresse: Instituttveien 20 2007 Kjeller

Telefon: 63 80 70 00 Telefaks: 63 80 71 15 Epost: post@ffi.no Norwegian Defence Research Establishment (FFI) P.O. Box 25 NO-2027 Kjeller

Office address: Instituttveien 20 N-2007 Kjeller

Telephone: +47 63 80 70 00 Telefax: +47 63 80 71 15 Email: post@ffi.no